UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): December 2, 2025

Mammoth Energy Services, Inc.

(Exact name of registrant as specified in its charter)

001-37917 (Commission File No.)

 Delaware
 32-0498321

 (State or other jurisdiction of incorporation or organization)
 (I.R.S. Employer Identification No.)

 14201 Caliber Drive, Suite 300

 Oklahoma City, Oklahoma
 (405) 608-6007
 73134

 (Address of principal executive offices)
 (Registrant's telephone number, including area code)
 (Zip Code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

Securities registered pursuant to Section 12(b) of The Act:

 Title of each class
 Trading Symbol(s)
 Name of each exchange on which registered

 Common Stock
 TUSK
 The Nasdaq Stock Market LLC

 NASDAQ Global Select Market
 NASDAQ Global Select Market

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§232.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging Growth Company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(s) of the Exchange Act.

[&]quot; Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

[&]quot; Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

[&]quot;Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

[&]quot;Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement.

Equity Purchase Agreement

On December 2, 2025, Mammoth Energy Partners LLC ("MEP"), a subsidiary of Mammoth Energy Services, Inc. ("Mammoth" or the "Company"), entered into an Equity Purchase Agreement (the "Agreement"), as the seller, with Qualus, LLC ("Qualus"), as the buyer, and Aquawolf LLC ("Aquawolf"), MEP's wholly-owned subsidiary and the subject of the sale, as a party to the Agreement. Pursuant to the Agreement, MEP sold all equity interests in Aquawolf, which was included in the Company's Infrastructure segment, to Qualus for \$30.0 million (the "Transaction"). The Transaction was completed simultaneously with the signing of the Agreement on December 2, 2025.

The aggregate sales price in connection with the Transaction was approximately \$30.0 million, subject to customary post-closing adjustments, including reductions for closing indebtedness, working capital shortfall, and transaction expenses. Of the \$30.0 million, \$23.5 million was paid to MEP and \$2.5 million was deposited into an escrow account for the purposes of funding post-closing adjustments for at least ninety days and indemnified liabilities until at least December 1, 2026. The Agreement includes customary representations, warranties and covenants by the parties. In addition, the Agreement provides for customary indemnification rights with respect to a breach of a representation, warranty or covenant by either party, subject to customary thresholds and caps on liability.

The foregoing summary does not purport to be complete and is qualified in its entirety by the full text of the Agreement, a copy of which is attached as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated by reference herein.

Revolving Credit Agreement Consent

In conjunction with the transaction described above, the Company entered into a consent and release agreement with Fifth Third Bank, National Association ("Fifth Third"), in which Fifth Third consented to the transaction and agreed to release associated collateral.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The information set forth above under Item 1.01 is hereby incorporated by reference into this Item 2.01.

Item 7.01 Regulation FD Disclosure.

On December 2, 2025, the Company issued a press release announcing the Transaction. A copy of the press release is furnished herewith as Exhibit 99.1 hereto and is incorporated by reference into this Item 7.01.

Item 9.01 Financial Statements and Exhibits.

(b) Unaudited Pro Forma Condensed Consolidated Financial Statements

As previously disclosed in the Company's Current Report on Form 8-K filed on April 17, 2025 and June 20, 2025, the Company completed the sale of its infrastructure services subsidiaries (the "T&D Transaction") and the sale of all of its hydraulic fracturing equipment and related assets (the "Pressure Pumping Transaction"), respectively. These prior transactions combined with ceasing operations of the Company's trucking and equipment manufacturing services, both of which were associated with pressure pumping and included within the Well Completion segment, reflect a strategic shift in the Company's operations and were classified as discontinued operations in the Company's financial statements, as reported in the Company's Quarterly Report on Form 10-Q for the periods ended June 30, 2025 and September 30, 2025.

The following Unaudited Pro Forma Condensed Consolidated Financial Statements of the Company reflecting (i) the transaction described in Item 1.01 of this Current Report on Form 8-K (ii) the previously disclosed T&D Transaction and Pressure Pumping Transaction and (iii) ceasing operations of the Company's trucking and equipment manufacturing services are filed as Exhibit 99.2 to this Current Report on Form 8-K and are incorporated by reference into this Item 9.01

- Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2025;
- Unaudited Pro Forma Condensed Consolidated Statements of Operations for the nine months ended September 30, 2025 and years ended December 31, 2024 and 2023; and
- Notes to Unaudited Pro Forma Condensed Consolidated Financial Statements.
- (d) Exhibits

Exhibit Number	Description
10.1*	Equity Purchase Agreement, dated as of December 2, 2025, by and among Aquawolf LLC, Mammoth Energy Partners LLC, Qualus LLC, and Mammoth Energy Services, Inc.
<u>99.1</u>	Press release dated December 2, 2025
<u>99.2</u>	Mammoth Energy Services, Inc. Unaudited Pro Forma Condensed Consolidated Financial Statements.
104	Cover Page Interactive Data File - the cover page XBRL tags are embedded within the Inline XBRL document.

^{*} Portions of the Equity Purchase Agreement and exhibits to the Equity Purchase Agreement have been omitted pursuant to Items 601(b)(2)(ii) and 601(a)(5) of Regulation S-K. The Company hereby undertakes to furnish copies of any of the omitted exhibits upon request by the SEC.

a.				
•	gn	o t		na
171	211	aı	u	

Date:

December 4, 2025

Pursuant to the requirements of the Securities Exchange Act of 1934	, the registrant has duly	caused this report to be s	signed on its behalf by the	undersigned thereunto duly
authorized.				

MAMMOTH ENERGY SERVICES, INC.

By: /s/ Mark Layton

Mark Layton

Chief Financial Officer and Secretary

[***] CERTAIN INFORMATION HAS BEEN EXCLUDED PURSUANT TO REGULATION S-K, ITEMS 601(B)(2)(II) AND 601(A)(5), FROM THIS DOCUMENT BECAUSE IT IS BOTH NOT MATERIAL AND IS THE TYPE THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL

EQUITY PURCHASE AGREEMENT

by and among

Aquawolf LLC,

Mammoth Energy Partners LLC,

Mammoth Energy Services, Inc.,

and

Qualus, LLC

Dated as of December 2, 2025

TABLE OF CONTENTS

		Page
ARTICLE I I	DEFINITIONS	1
1.1	Definitions	1
ARTICLE II	PURCHASE AND SALE OF THE PURCHASED EQUITY	13
2.1	Basic Transaction	
2.2	Closing Transactions	
2.3	Purchase Price	
2.4	Withholding	
ARTICI E III	REPRESENTATIONS AND WARRANTIES CONCERNING THE COMPAN	JV
ARTICLE III	SELLER AND SELLER MEMBER	200 CONTROL DE 200 C
3.1	Corporate Organization; Power and Authority; Legal Capacity	
3.2	Capitalization	
3.3	Authorization; Noncontravention	
3.4	Subsidiaries	
3.5	Financial Statements	
3.6	Absence of Undisclosed Liabilities	
3.7	No Material Adverse Effect	
3.8	Absence of Certain Developments.	
3.9	Assets	
3.10	Contracts and Commitments.	
3.11	Intellectual Property Rights	
3.12	Litigation	
3.13	Compliance with Laws	
3.14	Environmental, Health and Safety Matters	
3.15	Employees	
3.16	Employee Benefit Plans	
3.17	Insurance	
3.18	Tax Matters	
3.19	Brokerage [***]	
3.20	Affiliate Transactions	
3.21	Bank Accounts	
3.22	Anti-Bribery Compliance; International Trade	
3.23	Customer and Suppliers	
3.24	Backlog.	
3.25	Predecessor Entities	
3.23	Treaccessor Endices	
	REPRESENTATIONS AND WARRANTIES CONCERNING BUYER	
4.1	Organization and Power	
4.2	Authorization	
4.3	No Violation	
4.4	Governmental Authorities and Consents	
4.5	Litigation	
4.6	Brokerage	
4.7	Investment Purpose	42

ARTICLE V S	SURVIVAL; INDEMNIFICATION; OTHER COVENANTS	42
5.1	Survival of Representations and Warranties	
5.2	Indemnification	
5.3	Press Release and Announcements	46
5.4	Use of Name and Information	46
5.5	Expenses	46
5.6	Specific Performance	
5.7	Further Assurances	46
5.8	Confidentiality	
5.9	Tax Matters	
5.10	Release	49
5.11	Restrictive Covenants	50
5.12	Insurance Matters	51
ARTICLE VI	MISCELLANEOUS	
6.1	Amendment and Waiver	
6.2	Notices	
6.3	Successors and Assigns	
6.4	Severability	
6.5	Interpretation	
6.6	No Third Party Beneficiaries	54
6.7	Complete Agreement	54
6.8	Counterparts	54
6.9	Electronic Delivery	
6.10	Governing Law; JURISDICTION; WAIVER OF JURY TRIAL	
6.11	Schedules	55
6.12	Non-recourse	56
6.13	Time Periods	56

EXHIBITS

Exhibit A — Estimated Purchase Price Exhibit B — Accounting Principles

EQUITY PURCHASE AGREEMENT

THIS EQUITY PURCHASE AGREEMENT (this "Agreement") is entered into as of December 2, 2025, by and among (i) Aquawolf LLC, a Delaware limited liability company (the "Company"), (ii) Mammoth Energy Partners LLC, a Delaware limited liability company ("Seller"), (iii) Mammoth Energy Services, Inc., a Delaware corporation ("Seller Member"), and (iii) Qualus, LLC, a Florida limited liability company ("Buyer").

WHEREAS, the Company provides a range of specialized engineering and consulting solutions to clients in the electrical transmission and distribution as well as the natural gas utility, infrastructure, and related industries, including utility engineering and design, owner's engineer services, project management, scheduling, utility records research plotting, and consulting for electrical transmission and distribution as well as for natural gas pipelines, performance of technical design and procurement reviews, offering quality assurance and quality control services, third-party antenna attachment analysis (which for the avoidance of doubt, excludes wireless telecommunication carriers as well as tower and infrastructure real estate investment trusts and other entities that own and lease tower assets), system modeling and analysis, feasibility studies, greenfield designs, brownfield retrofits and upgrades, and substation and switchyard design (the "Business");

WHEREAS, Seller Member owns all of the issued and outstanding Equity Interests of and in Seller;

WHEREAS, Seller owns all of the issued and outstanding Equity Interests (as defined herein) of and in the Company (the "Purchased Equity");

WHEREAS, on the terms set forth in this Agreement, Buyer desires to purchase from Seller, and Seller desires to sell to Buyer, all of the Purchased Equity; and

[***]

NOW, THEREFORE, in consideration of the mutual covenants, agreements and understandings contained herein and intending to be legally bound, the parties hereto hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. For purposes of this Agreement, the following terms have the meanings set forth below:

"Action" means any action, claim, suit (whether civil, criminal, administrative or judicial), charge, complaint, cause of action, demand, lawsuit, mediation, arbitration, audit, inquiry, proceeding (including any administrative enforcement proceeding, arbitration or mediation proceeding or proceeding in eminent domain), controversy, litigation, summons, subpoena or any civil, criminal, administrative or regulatory investigation.

"ACA" shall have the meaning set forth in Section 3.16(c).

"Adjustment Calculation Time" means 12:01 a.m. prevailing Central Daylight Time on the Closing Date; provided that, with respect to any calculation made pursuant to this Agreement as of the Adjustment Calculation Time, such calculation shall factor in and include the use or transfer of any assets of the Company (including Cash) (a) to repay or discharge any Indebtedness or pay any Transaction Expenses, or

(b) outside of the ordinary course of business (including to the extent such assets are sold, liquidated, disposed of or otherwise used to pay any distributions to equityholders of the Company or repurchase any Equity Interests of the Company), in each case following the Adjustment Calculation Time and prior to the Closing.

"Adjustment Escrow Amount" means Two Hundred Thousand Dollars (\$200,000).

"Affiliate" of any particular Person means any other Person controlling, controlled by or under common control with such Person, and, in the case of an individual, any individual related by blood, marriage or adoption to any such individual or any Person in which any such individual owns any beneficial interest or the estate of such individual, and, in the case of any trust, any grantor(s) or beneficiary(ies) of such trust. For purposes of this definition, "control" (including the terms "controlling," "controlled by" and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise, and such "control" will be presumed if any Person owns ten percent (10%) or more of the voting capital stock or other ownership interests, directly or indirectly, of any other Person; provided, however, that, Sponsor and its Subsidiaries and portfolio companies (other than Seller Member and its Subsidiaries) shall not be an Affiliate of the Company for purposes of Article III ((other than Section 3.20 (Affiliate Transactions)) or Section 5.11 hereof.

"Affiliated Group" means any affiliated group as defined in Section 1504 of the Code (or any analogous combined, consolidated or unitary group defined under state, local or foreign income Tax law) of which the Company is or has been a member.

"Agreement" shall have the meaning set forth in the preamble.

"AI Tools" mean Generative AI platforms or tools used in connection with the operation of the businesses of the Company.

"Allocation Statement" shall have the meaning set forth in Section 5.9(h).

"Backlog" shall have the meaning set forth in Section 3.24.

"Base Purchase Price" means Thirty Million Dollars (\$30,000,000).

"Business" shall have the meaning set forth in the recitals.

"Business Data" means all business information and all Personal Information and data (whether of employees, contractors, consultants, customers, consumers, or other Persons and whether in electronic or any other form or medium) that is accessed, collected, used, processed, stored, shared, distributed, transferred, disclosed, destroyed, or disposed of by the Company.

"Business Day" means a day on which commercial banks in the United States of America are open for the transaction of commercial banking business, which will not include a Saturday, Sunday, or national holiday.

"Buyer" shall have the meaning set forth in the preamble.

"Buyer Indemnitees" means the Company, Buyer and their respective Affiliates and their respective equityholders, officers, managers, directors, employees, agents, partners, members, representatives, successors and assigns.

"CARES Act" means (a) the Coronavirus Aid, Relief, and Economic Security Act (Pub. L. 116-136) and any administrative or other guidance published with respect thereto by any Governmental Authority, or any other Law intended to address the consequences of COVID-19, and (b) any extension of, amendment, supplement, correction, revision or similar treatment to any provision of the CARES Act contained in the Consolidated Appropriations Act, 2021, H.R. 133.

"Cash" means, at any given time, without duplication, the actual cash balances of the Company and deposits in transit but not yet credited, each as determined in accordance with GAAP, less (a) any restricted or trapped cash, (b) any bank overdrafts, (c) any outstanding checks, wires, transfers and drafts, (c) any deposits of third parties and any cash held as collateral and (d) any cash and cash equivalents received in connection with insurance claim(s) to the extent the underlying damage, destruction or loss giving rise thereto has not been cured or remediated. For the avoidance of doubt, "Cash" can be a negative number.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601, et seq.), as amended, and all rules, regulations and standards issued thereunder.

[***]

"Closing" shall have the meaning set forth in Section 2.2(a).

"Closing Cash" means the aggregate amount of Cash as of the Adjustment Calculation Time. For the avoidance of doubt, "Closing Cash" can be a negative number.

"Closing Date" shall have the meaning set forth in Section 2.2(a).

"Closing Indebtedness" means the aggregate amount of all Indebtedness of the Company as of immediately prior to the Closing.

"Closing Net Working Capital" means Net Working Capital as of the Adjustment Calculation Time.

"Closing Statement" shall have the meaning set forth in Section 2.3(b).

"Closing Statement Delivery Deadline" shall have the meaning set forth in Section 2.3(b).

"COBRA" means Part 6 of Subtitle B of Title I of ERISA, Section 4980B of the Code and any similar state law.

"Code" means the Internal Revenue Code of 1986, as amended, and any reference to any particular Code section shall be interpreted to include any revision of or successor to that section regardless of how numbered or classified.

"Company" shall have the meaning set forth in the preamble.

"Company AI Products" mean all Company Products that employ, deploy, leverage, or make use of Generative AI.

"Company Intellectual Property" shall have the meaning set forth in Section 3.11(b).

"Company Owned Intellectual Property" shall have the meaning set forth in Section 3.11(f).

"Company Products" means all Software and other products, including any of the foregoing currently in development, from which the Company has derived, is currently deriving or is scheduled to derive, revenue from the sale, license, maintenance or other provision thereof.

"Confidential Information" means all information, in any form or medium, that relates (a) to the Company, including the business, products, financial position or condition, services or research or development of the Company or (b) to the respective suppliers and vendors, distributors, referral sources, customers, independent contractors or other business relations of the Company. Confidential Information shall not include any information that is or becomes generally known to, or available for use by, the public other than as a result of any acts or omissions of Seller, Seller Member, Sponsor or any of their respective Affiliates or Representatives.

"Covered Tax Proceeding" shall have the meaning set forth in Section 5.9(e).

"COVID-19" means the coronavirus disease known as SARS CoV2 or COVID-19, and any evolutions or mutations thereof or associated epidemic, pandemic or disease outbreaks.

"Data Security Requirements" means, collectively, all of the following to the extent relating to Processing or otherwise relating to privacy, security, or security breach notification requirements and applicable to the Company, to the conduct of the Company's business, or to any of the Systems or any Business Data: (a) the Company's own rules, policies, guidelines, and procedures, including but not limited to, Company's internal and externally posted privacy policies or notices; (b) all applicable laws, rules and regulations; (c) industry standards applicable to the industry in which the Company operates (including, if applicable, the Payment Card Industry Data Security Standard (PCI DSS)); and (d) contracts into which the Company has entered or by which it is otherwise bound.

"Encumbrance" means any Lien (other than restrictions on transfer under the Securities Act and applicable state securities laws).

"Enforceability Limitations" shall have the meaning set forth in Section 3.3(a).

"Environmental and Safety Requirements" means all Laws concerning, related to, or involving in any way public health and safety, worker health and safety, exposure to Hazardous Materials, pollution or protection of the environment, including all those relating to the presence, use, production, generation, handling, transport, treatment, storage, disposal, distribution, labeling, testing, processing, discharge, release, threatened release, control or cleanup of, or exposure to, any Hazardous Materials, including CERCLA and RCRA.

"Equity Interests" means any and all stock, shares, interests, participations or other equivalents (however designated) of capital stock of a corporation and any and all ownership interests in a Person other than a corporation, including units, membership interests, partnership interests, joint venture interests and beneficial interests, and any and all warrants, options or other rights to purchase or other acquire any of the foregoing.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means any Person (whether or not incorporated) that, together with the Company, could be, or at any relevant time could have been, treated as a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA.

"Escrow Agent" means [***].

"Escrow Agreement" means the Escrow Agreement, by and among Seller, Buyer and the Escrow Agent governing the administration of the Adjustment Escrow Amount and the Indemnification Escrow Amount.

"Escrow Amount" means the Indemnification Escrow Amount plus the Adjustment Escrow Amount.

"Estimated Purchase Price" means the amount representing Seller's good faith estimate, as of the date hereof, of the Purchase Price, which such amount and the calculation and components thereof are set forth on Exhibit A hereto.

"Finalization Date" shall have the meaning set forth in Section 2.3(e).

"Final Purchase Price" shall have the meaning set forth in Section 2.3(e).

"Financial Statements" shall have the meaning set forth in Section 3.5(a).

"Fundamental Representations" means the representations and warranties contained in Section 3.1 (Corporate Organization; Power and Authority; Legal Capacity), Section 3.2 (Capitalization), Section 3.3 (Authorization; Noncontravention) (other than the representations and warranties in Section 3.3(b) with respect to Material Contracts and Permits), Section 3.4 (No Subsidiaries), Section 3.9(a) (Assets), Section 3.19 (Brokerage [***]) and Section 3.20 (Affiliate Transactions).

"GAAP" means United States generally accepted accounting principles in effect from time to time.

"Generative AI" means artificial intelligence tools or other technology capable of generating various types of content including text, code, data, images, or other media, using models that have been trained on data sources for the purposes of generating output with any characteristics similar to such data.

"Governing Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of incorporation, articles of association, formation or organization and any joint venture, limited liability company, operating or partnership agreement and other similar documents adopted or filed in connection with the creation, formation or organization of each such Person and (b) all bylaws, voting agreements, stockholder agreements, investor rights agreements and similar documents, instruments or agreements relating, in each case, to the organization or governance of such Person, or the transfer of securities of such Person, in each case, as amended, restated, supplemented or otherwise modified.

"Governmental Authority" means the United States of America or any other nation, any state or other political subdivision thereof, or any entity or Person exercising executive, legislative, judicial, regulatory or administrative functions of government or any agency or department or subdivision of any governmental authority, including the United States federal government or any state or local government, or any arbitrator or other arbitral body (public or private).

"Guaranty" means any agreement, undertaking or arrangement by which any Person guarantees, endorses or otherwise becomes or is contingently liable for the debt, obligation or other liability of any other Person (other than by endorsements of instruments in the ordinary course of collection).

"Hazardous Material" means any waste, substance, chemical, compound, or material listed under, regulated by, or which may give rise to liability or standards of conduct pursuant to, Environmental and Safety Requirements, including hazardous substances, pesticides, pollutants, contaminants, toxic chemicals, petroleum products or byproducts, asbestos or asbestos-containing material, polychlorinated

biphenyls, contaminants of emerging concern, including per and polyfluoroalkyl substances, noise, odor, mold, radiation, radon, and any and all other substances or materials that are or may be reasonably determined to present a threat, hazard or risk to human health or the environment and is or may become regulated by or under Environmental and Safety Requirements.

"Income Tax" means Taxes imposed on the Company at the entity level based on net income, any non-resident withholding Taxes required to be paid by the Company or on behalf of Seller (or any of Seller's Affiliates), and any Taxes as a result of a SALT Election.

"Indebtedness" means, with respect to any Person at any date, without duplication, (a) all obligations of such Person for borrowed money or in respect of loans or advances, (b) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments or debt securities of any kind or nature, (c) all obligations in respect of letters of credit, surety bonds, performance bonds and bankers' acceptances or similar arrangements issued for the account of such Person (including any Affiliated loans), (d) all obligations arising out or in respect of outstanding severance obligations and unpaid bonuses or commissions, (e) all obligations required to be recorded as capital or finance leases in accordance with GAAP as of the date of determination of such Indebtedness or any obligations related to leases that are recorded as capital or finance leases in the Financial Statements (disregarding for purposes of this clause the effect of ASC 842), (f) all unpaid income Taxes of such Person in respect of any Pre-Closing Tax Period (including the portion of any Straddle Period ending on or before the Closing Date), regardless of whether then due and owing, which shall not be less than zero in any jurisdiction and shall not include any offsets or reductions with respect to Tax refunds, but which shall take into account any payments of estimated Tax to the extent made prior to the Closing Date and which would reduce any such unpaid income Taxes in the relevant jurisdiction for the relevant taxable period, (g) all unpaid "applicable employment taxes" or other payroll taxes deferred pursuant to the CARES Act or the Payroll Tax Executive Order, (h) any self-insured liabilities for worker's compensation, health and welfare obligations or other matters, (i) all obligations (net of any portion of said obligations that are in accounts receivable) for any rebates, overpayments, billing and fee reconciliations and demands therefor owing to third parties, (j) declared but unpaid dividends or distributions, (k) all obligations due to equityholders or Affiliates of such Person or payable to third parties for the benefit of any Affiliates of such Person (including, for the avoidance of doubt, any accrued and unpaid property taxes), (1) all financed insurance premiums with respect to the Insurance Policies, and (m) all obligations in respect of accrued and unpaid interest on and any fees, prepayment premiums, penalties make-whole payments and other similar fees, expenses or payments related to any of the foregoing.

"Indemnification Escrow Amount" means Two Million Two Hundred and Fifty Thousand Dollars (\$2,250,000).

"Indemnification Escrow Release Date" shall have the meaning set forth in Section 5.2(c).

"Indemnified Taxes" means (a) any and all Taxes (or the nonpayment thereof) of the Company (including any Taxes required to be collected through withholding) for all Pre-Closing Tax Periods, (b) any and all Taxes of any Affiliated Group of which the Company is or was a member on or prior to the Closing Date, including pursuant to Treasury Regulation Section 1.1502-6 or any analogous or similar state, local, or non-U.S. Law or regulation, (c) any and all Taxes of any Person (other than the Company) imposed on the Company as a transferee or successor, by contract or pursuant to any Law, rule or regulation or otherwise, which Taxes relate to an event or transaction occurring before the Closing, and (d) reasonable out-of-pocket and third-party costs and expenses associated with preparing, filing, re-filing, amending, revoking, correcting, or modifying any Tax Return with respect to any Pre-Closing Tax Period (including, for the avoidance of doubt, such costs and expenses of any Tax Return allocable to the portion of any Straddle Period ending on the Closing Date).

"Indemnitee" shall have the meaning set forth in Section 5.2(d).

"Indemnitor" shall have the meaning set forth in Section 5.2(d).

"Independent Accountant" means [***], or if [***] refuses or is otherwise unable to act as the Independent Accountant then Buyer and Seller shall cooperate in good faith to appoint a nationally recognized accounting firm that is mutually agreed upon by Buyer and Seller in writing to serve as the Independent Accountant; provided, that if Buyer and Seller are unable to agree on such a firm, they shall each nominate such a firm, and the two firms nominated shall nominate a third such firm, with such third firm to serve as the Independent Accountant.

"Insurance Claim" shall have the meaning set forth in Section 5.12.

"Insurance Policies" shall have the meaning set forth in Section 3.17.

"Intellectual Property Rights" means any and all intellectual and industrial proprietary and property rights anywhere in the world, including the following: (a) patents, patent applications, patent disclosures, invention disclosures and inventions (regardless of whether patentable and regardless of whether reduced to practice) and any reissue, continuation, continuation-in-part, division, revision, extension or reexamination thereof; (b) industrial design rights (including utility model rights, design rights, and industrial property rights) and registrations and applications for registration of industrial design rights; (c) Internet domain names, trademarks, service marks, trade dress, trade names, logos, slogans, company names, corporate names, phone numbers containing any of the foregoing (and all translations, adaptations, derivations and combinations of the foregoing), and registrations and applications for registration thereof together with all of the goodwill associated therewith; (d) copyrights (registered or unregistered) and copyrightable works, and registrations and applications for registration thereof; (e) mask works and registrations and applications for registration thereof; (f) rights in Software (in both source code and object code form) and documentation thereof; (g) trade secrets and confidential and proprietary information (including ideas, formulas, recipes, compositions, knowhow, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, data and databases, personally identifiable information, financial and marketing plans and customer and supplier lists and information); and (h) rights of privacy, publicity, and endorsement (including rights to the use of names, voices, likenesses, images, appearances, signatures, and biographical information of natural persons).

"Inventory" means all raw materials, work-in-process, finished goods, supplies, parts, and packaging materials sold, used, or consumed by the Company, whether (a) located at a facility of the Company; (b) in transit to a facility of the Company, or (c) at a location separate from the facility of the Company.

"Investment" as applied to any Person means (a) any direct or indirect purchase or other acquisition by such Person of any notes, obligations, instruments, stock, securities or ownership interest (including limited liability company interests, partnership interests and joint venture interests) of any other Person, and (b) any capital contribution by such Person to any other Person.

"Key Employees" means, collectively, [***], [***], [***] and [***].

"Knowledge" means, when referring to the "Knowledge" of Seller or any other similar knowledge qualification, the actual knowledge of each of the Key Employees and [***] after due inquiry with their direct reports who would reasonably be likely to have knowledge of the particular matter.

"Labor Agreement" shall have the meaning set forth in Section 3.10(a)(v).

"Latest Balance Sheet" shall have the meaning set forth in Section 3.5(a).

"Latest Balance Sheet Date" shall have the meaning set forth in Section 3.5(a).

"Law" means any U.S. or non-U.S. federal, state, local or municipal statute, act, law, ordinance, regulation, directive, rule, code, order, judgment, injunction, ruling, award, writ, guidance, decree or principle of common law enacted, promulgated, adopted, applied, issued, enforced or entered by any Governmental Authority.

"Lease" and "Leases" shall have the meaning set forth in Section 3.9(c).

"Leased Real Property" shall have the meaning set forth in Section 3.9(c).

"Lien" means any mortgage, pledge, hypothecation, deed of trust, lien, security interest, charge, claim, encroachment, community property interest, security agreement, easement, covenant, restriction, option, license (or sublicense) or other encumbrance or restriction of any kind or nature whatsoever.

"Loss" means any loss, liability, demand, judgment, claim, action, cause of action, cost, damage, Tax, penalty, fine or expense, whether or not arising out of third party claims (including interest, penalties, reasonable legal, consulting and other professional fees and expenses and all amounts paid in prosecution, investigation, defense or settlement of any of the foregoing).

"Malicious Code" means any virus, worm, Trojan horse or similar disabling code or program.

"Material Adverse Effect" means a material and adverse effect or development upon (a) the ability of Seller, Seller Member or the Company to execute or deliver this Agreement, to perform any of their obligations under this Agreement or to consummate any of the transactions contemplated by this Agreement or the ability of Buyer to receive the full benefit of the transactions contemplated by this Agreement or (b) the business, operations, assets, liabilities, financial condition or position, operating results, cash flow, net worth, supplier or other business relations of the Company taken as a whole; provided, however, that solely with respect to the foregoing clause (b), "Material Adverse Effect" shall not include any effect or development to the extent arising out of or attributable to: (i) general economic or political conditions; (ii) conditions generally affecting the industries in which the Company operates; (iii) any changes in financial, banking or securities markets in general, including any disruption thereof and any decline in the price of any security or any market index or any change in prevailing interest rates; (iv) acts of war (whether or not declared), armed hostilities or terrorism, or the escalation or worsening thereof; or (v) any natural or manmade disaster or acts of God, in each case, except to the extent that any such matters disproportionately impact the Company, taken as a whole, relative to the other businesses in the industries in which the Company operates.

"Material Contract" means the contracts, leases, licenses (or sublicenses), agreements and instruments set forth, or required to be set forth, on Schedule 3.10.

"Material Customers" shall have the meaning set forth in Section 3.23(a).

"Material Vendors" shall have the meaning set forth in Section 3.23(a).

"Materiality Qualifier" means any qualification referencing the terms "material," "materially," "materiality," "Material Adverse Effect," "material adverse change," "material respects," "except where

the failure to so comply would not reasonably be expected to have a Material Adverse Effect," "for such exceptions that would not reasonably be expected to have a Material Adverse Effect" or other any similar terms, clauses or phrases of similar import or effect, other than the terms "Material Contracts" and "Material Vendors."

"Net Working Capital" means (x) the total current assets of the Company minus (y) the total current liabilities of the Company, in each case calculated in accordance with GAAP and the methodology set forth on Exhibit B (the "Accounting Principles"). To the extent the sample calculation set forth in Exhibit B-1 conflicts with GAAP, GAAP shall prevail; provided, however, that the illustrative adjustments to the rebate due to [***] (as offset by receivables related to such) set forth in Exhibit B-1 shall apply rather than GAAP such that Closing Net Working Capital shall be increased by the net amount of rebates due to [***].

"Net Working Capital Target" means One Million Two Hundred and Eighty Thousand Dollars (\$1,280,000), as of the Adjustment Calculation Time.

"Non-Compete Restricted Period" means a period of three (3) years from the Closing Date.

"Non-Recourse Parties" shall have the meaning set forth in Section 6.12(a).

"Non-Solicit Restricted Period" means a period of five (5) years from the Closing Date.

"Notice of Disagreement" shall have the meaning set forth in Section 2.3(c).

"Open Source Software" means any Software that is licensed pursuant to: (a) any license that is a license now or in the future approved by the Open Source Initiative and listed at http://www.opensource.org/licenses, which licenses include all versions of the GNU General Public License (GPL), the GNU Lesser General Public License (LGPL), the GNU Affero GPL, the MIT license, the Eclipse Public License, the Common Public License, the CDDL, the Mozilla Public License (MPL), the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL), and the Sun Industry Standards License (SISL); (b) any license to Software that is considered "free" or "open source software" by the Open Source Foundation or the Free Software Foundation; or (c) any Reciprocal License, in each case whether or not source code is available or included in such license.

"Parent" shall have the meaning set forth in the recitals.

"Pass-Through Tax Return" means any federal, state, local or non-U.S. Tax Returns that report income with respect to the Company but with respect to which the direct or indirect beneficial owners of the Company are required to pay the related Tax.

"Payroll Tax Executive Order" means the Presidential Memorandum on Deferring Payroll Tax Obligations in Light of the Ongoing COVID-19 Disaster, as issued on August 8, 2020 and including any U.S. presidential memorandum, executive order or similar pronouncement permitting or requiring the deferral of any payroll Taxes (including those imposed by Section 3101(a) and 3201 of the Code, IRS Notice 2020-65 and IRS Notice 2021-11).

"Permits" means all licenses, permits, registrations, accreditations, certifications, qualifications, approvals and consents pending with or issued by Governmental Authorities or accreditation organizations, or the administrative agents thereof.

"Permitted Liens" means (a) Liens for Taxes not due and payable, for which adequate reserves have been maintained on the Latest Balance Sheet in accordance with GAAP, (b) statutory landlord's, mechanic's, carrier's, workmen's, repairmen's or other similar Liens arising or incurred in the ordinary course of business for amounts which are not due and payable, (c) Liens arising from zoning ordinances which are not material to the business of the Company as currently conducted thereon, and (d) with respect to Liens on Intellectual Property Rights, nonexclusive licenses (or sublicenses) granted to customers in the ordinary course of business.

"Person" means any individual, sole proprietorship, partnership, joint venture, trust, unincorporated association, corporation, limited liability company, other entity or Governmental Authority.

"Personal Information" means any and all information, whether alone or together with any other data or information, (i) relating to, describing, is reasonably capable of being associated with or could be reasonably linked with an identified or identifiable individual or household, or (ii) information that constitutes "personal information" or "personal data" or "protected health information" or any similar term under applicable Data Security Requirements.

"Plan" shall have the meaning set forth in Section 3.16(a).

"PPP Loan" means any "Paycheck Protection Program" loan(s), grant(s) or similar financial assistance in connection with the CARES Act.

"Pre-Closing Tax Period" means taxable period ending on or before the Closing Date and the portion through the end of the Closing Date for any Straddle Period.

"Proceeding" shall have the meaning set forth in Section 3.12.

"Processing" means the access, collection, use, processing, storage, sharing, adaptation or alteration, retrieval, distribution, transmission, dissemination or otherwise making available, transfer, disclosure, security, destruction, or disposal of any personal, sensitive, or confidential information or data (whether in electronic or any other form or medium).

"Purchased Equity" shall have the meaning set forth in the recitals.

"Purchase Price" shall have the meaning set forth in Section 2.2(a).

"RCRA" means the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.), as amended, and all rules, regulations and standards issued thereunder.

"Reciprocal License" means a license of an item of Software that requires or that conditions any rights granted in such license upon: (a) the disclosure, distribution or licensing of any other Software (other than such item of Software as provided by a third party in its unmodified form); (b) a requirement that any disclosure, distribution or licensing of any other Software (other than such item of Software in its unmodified form) be at no charge; (c) a requirement that any other licensee of the Software be permitted to access the source code of, modify, make derivative works of, or reverse-engineer any such other Software; (d) a requirement that such other Software be redistributable by other licensees; or (e) the grant of any patent rights (other than patent rights in such item of Software), including non-assertion or patent license obligations (other than patent obligations relating to the use of such item of Software).

"Release" means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, placing, discarding, abandonment, or disposing into

the environment (including the placing, discarding or abandonment of any barrel, container or other receptacle containing any Hazardous Material).

"Released Claims" shall have the meaning set forth in Section 5.10.

"Releasors" shall have the meaning set forth in Section 5.10.

"Releasees" shall have the meaning set forth in Section 5.10.

"Registered Company Intellectual Property" shall have the meaning set forth in Section 3.11(a).

"Representatives" means, with respect to any Person, any and all directors, managers, officers, employees, consultants, financial advisors, counsel, accountants and other agents of such Person.

"Restricted Party" and "Restricted Parties" shall have the meanings set forth in Section 5.11(a).

"SALT Election" means an election under applicable state or local Income Tax Laws made by the Company pursuant to which the Company incurs or is otherwise liable for any state or local Tax liability under applicable state or local Income Tax Law that would have been borne (in whole or in part) by the direct or indirect equity owners of the Company had no such election been made (e.g., any "Specified Income Tax Payment" as defined by IRS Notice 2020-75).

"Securities Act" means the Securities Act of 1933, as amended, or any similar federal law then in force.

"Security Incident" means actions that result in an actual, suspected, alleged or potentially likely cyber or security incident that could have an adverse effect on any Systems, Confidential Information or Business Data, including an occurrence that actually or potentially likely jeopardizes the confidentiality, integrity, or availability of any Systems, Confidential Information or Business Data. A Security Incident includes incidents of security breaches or intrusions, denial of service, or unauthorized entry, access, collection, use, processing, storage, sharing, distribution, transfer, disclosure, or destruction of any Systems, Confidential Information or Business Data, or any loss, distribution, compromise or unauthorized access to or disclosure of any of the foregoing.

"Seller" shall have the meaning set forth in the preamble.

[***]

[***]

"Seller Indemnitees" means Seller and its Affiliates and their respective equityholders, officers, managers, directors, employees, agents, partners, members, representatives, successors and assigns.

"Seller Member" shall have the meaning set forth in the preamble.

"Seller Member Insurance Policies" shall have the meaning set forth in Section 5.12.

"Software" means computer software (in object code or source code format), computer programs, applications, specifications, interfaces, definitions, utilities, development tools, diagnostics, and embedded systems, in any form or medium, including source code, object code and executable code, and all databases and data used with, or used to develop, any of the foregoing, together with all related user manuals, programmer documentation, text, diagrams, graphs, charts, and other documentation and materials.

"Sponsor Restrictive Covenants Agreement" means the Restrictive Covenants Agreement, dated as of the date hereof, by and among Wexford Capital LP, a Delaware limited partnership ("Sponsor"), and Buyer.

"Straddle Period" shall have the meaning set forth in Section 5.9(a).

"Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association, or business entity of which (a) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (b) if a limited liability company, partnership, association, or other business entity (other than a corporation), a majority of partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person shall be deemed to have a majority ownership interest in a limited liability company, partnership, association, or other business entity (other than a corporation) if such Person shall be allocated a majority of limited liability company, partnership, association, or other business entity gains or losses or shall be or control any managing member or general partner of such limited liability company, partnership, association or other business entity.

"Systems" means any and all Software, computer hardware (whether general purpose or special purpose), electronic data processing, information, record keeping, communications, telecommunications, networks, peripherals and computer systems, and other information technology systems, platforms and processes, including any outsourced systems, platforms and processes, and other similar or related items of automated, computerized and/or Software systems, platforms and processes that are owned, used or held for use by or on behalf of Company, including all data and information stored in or transmitted by any of the foregoing and documentation related to any of the foregoing.

"Tail Policies" means a prepaid and non-cancellable customary directors' and officers' liability, employment practices liability, and fiduciary liability insurance "tail policy" for the Company with a six-year extended reporting period from the Closing Date.

"Tax" means any federal, state, local or foreign income, gross receipts, franchise, profits, escheat, abandoned or unclaimed property (whether or not designated as a tax under applicable Law), estimated, alternative minimum, addon minimum, sales, use, transfer, registration, value added, excise, natural resources, severance, stamp, occupation, premium, windfall profit, environmental, customs, duties, real property, personal property, capital stock, social security, employment, unemployment, disability, payroll, license, employee or other withholding, privilege, duty, ad valorem or other tax, of any kind whatsoever and however denominated, including any interest, penalties or additions to tax or additional amounts in respect of the foregoing, regardless of whether disputed.

"Tax Returns" means returns, elections, declarations, reports, claims for refund, information returns or other documents (including any related or supporting schedules, statements, attachments or information) filed with or submitted to, or required to be filed or submitted in connection with the determination, assessment or collection of any Taxes of any Person or the administration of any Laws, regulations or administrative requirements relating to any Taxes, and including any amendments thereof.

"Training Data" means training data, validation data, test data, scraped or harvested datasets or databases, in each case used to train, finetune, enhance or improve Generative AI.

"Training Dataset" means any set of Training Data that the Company has used in the development, ongoing operation or improvement of any Generative AI and the license terms applicable thereto.

[***]

"Transaction Documents" means, collectively, this Agreement and all of the schedules, exhibits, certificates, agreements and other documents contemplated thereby and delivered in connection herewith, including the Transition Services Agreement and the Escrow Agreement.

"Transaction Expenses" means all unpaid fees, costs and expenses (including investment bankers', attorney's and accountant's fees, costs and expenses) incurred by or on behalf of (or that are otherwise payable by) Seller, the Company and/or Seller Member in connection with the transactions contemplated by this Agreement and the other Transaction Documents (or any alternative transaction hereto), whenever due and payable, including, for the avoidance of doubt, (a) any brokerage fees, commissions, finder's fees or financial advisory fees (in each case including all related costs and expenses), (b) [***], (c) fifty percent (50%) of the costs and expenses associated with the Tail Policies, (d) one hundred (100%) of the fees and expenses with respect to the Escrow Agent, and (e) Seller's portion of any Transfer Taxes pursuant to Section 5.9(i).

"Transfer Taxes" shall have the meaning set forth in Section 5.9(i).

"Transition Services Agreement" means the Transition Services Agreement, dated as of the date hereof, by and among Seller, the Company and Buyer.

"WARN Act" means the Worker Adjustment Retraining and Notification Act of 1988 as amended, 29 U.S.C. § 2101 et seq. and any similar Law.

ARTICLE II

PURCHASE AND SALE OF THE PURCHASED EQUITY

2.1 <u>Basic Transaction</u>. On the terms set forth in this Agreement, at the Closing, Buyer shall and hereby does purchase from Seller, and Seller shall and hereby does sell, convey, assign, transfer and deliver to Buyer, all of Seller's right, title and interest in and to the Purchased Equity, free and clear of all Encumbrances.

2.2 Closing Transactions.

- (a) The purchase and sale of the Purchased Equity contemplated by this Agreement (the "Closing") shall take place on the date hereof (the "Closing Date") remotely via the electronic (i.e., email/PDF) delivery of documents and signatures on the date hereof. The Closing will be deemed to be effective as of 12:01 a.m. Central Daylight Time on the Closing Date. The aggregate purchase price for the Purchased Equity (the "Purchase Price") payable by Buyer to Seller on the Closing Date shall be an aggregate amount equal to the result of (i) the Base Purchase Price, plus (ii) Closing Cash, minus (iii) Closing Indebtedness, plus (iv) the aggregate amount (if any) by which Closing Net Working Capital is greater than the Net Working Capital Target, minus (v) the aggregate amount (if any) by which Closing Net Working Capital is less than the Net Working Capital Target, minus (vi) the aggregate amount of all Transaction Expenses.
 - (b) At the Closing, on the terms set forth in this Agreement:

- (i) Buyer shall pay, or cause to be paid, to Seller an amount equal to the Estimated Purchase Price less the Escrow Amount by wire transfer of immediately available funds to the account designated by Seller prior to the date hereof;
- (ii) Buyer shall pay, or cause to be paid, to the beneficiaries thereof (as identified by Seller to Buyer) (A) all Indebtedness for borrowed money of the Company outstanding as of the Closing as set forth on Schedule 2.2(b)(ii)(A), and (B) the Transaction Expenses due and owing as of the Closing set forth on Schedule 2.2(b)(ii)(B); [***], if any, shall be paid or caused to be paid to the Company for further distribution to the beneficiaries thereof (as identified by Seller to Buyer);
- (iii) Buyer shall pay, or cause to be paid, to the Escrow Agent, the Escrow Amount;
 - (iv) Buyer shall deliver to Seller, the following:
 - (A) the Transition Services Agreement, duly executed by Buyer;
 - (B) the Escrow Agreement, duly executed by Buyer, and
 - (C) the Sponsor Restrictive Covenants Agreement, duly executed by

Buyer;

- (v) Seller shall deliver to Buyer the following:
- (A) good and valid title to all of the Purchased Equity, free and clear of all Encumbrances, together with duly executed instruments evidencing the assignment of the Purchased Equity;
- (B) evidence that all of the agreements and arrangements set forth on Schedule 2.2(b)(v)(B) have been terminated and are of no further force or effect without any further liabilities of the Company thereunder or in connection therewith;
- (C) evidence of releases of all Liens (other than any Permitted Liens) related to the assets and properties of the Company and payoff letters with respect to any Indebtedness for borrowed money of the Company outstanding as of the Closing as set forth on Schedule 2.2(b)(ii)(A), and terminations of guarantees or other obligations thereunder, associated with the related Indebtedness incurred or guaranteed by the Company or otherwise related to the Purchased Equity, and confirmation that the holder of such Indebtedness will release such Liens and terminate such guarantees or obligations, in each case, in a form reasonably acceptable to Buyer;
- (D) all third-party consents and approvals set forth on Schedule 2.2(b)(v)(D);
- (E) all governmental and regulatory consents, novations, approvals, licenses and authorizations set forth on Schedule 2.2(b)(v)(E);
- (F) copies of the resolution of the Company's member authorizing the execution, delivery and performance of this Agreement and the other Transaction Documents and the consummation of the transactions contemplated hereby and thereby;

- (G) the resignation, effective as of the Closing, of each officer, director, and manager (in their capacity as such), of the Company;
- (H) a good standing certificate for the Company from its jurisdiction of organization and each jurisdiction in which the Company is qualified to do business as a foreign entity, in each case, dated as of within five (5) Business Days prior to the Closing Date;
- (I) a properly completed and duly executed IRS Form W-9 from Seller;
- (J) invoices with respect to all Transaction Expenses (other than with respect to any Transaction Expenses that are compensatory payments to employees or other service providers of the Company);
- (K) all books and records and other property of the Company in Seller's possession or under Seller's control;
- (L) evidence that Buyer and Buyer's Affiliates (as designated by Buyer) have been added as authorized Persons to the bank accounts set forth on <u>Schedule 3.21</u> and the removal of [***] and [***] as authorized Persons from such bank accounts;
- (M) the Transition Services Agreement, duly executed by Seller and the Company;
 - (N) the Sponsor Restrictive Covenants Agreement, duly executed by

Sponsor;

(O) the Escrow Agreement, duly executed by Seller and the Escrow

Agent;

- (P) [***]
- (O) [***]

2.3 Purchase Price.

- (a) The aggregate purchase price for the Purchased Equity payable by Buyer to Seller on the Closing Date shall be the Purchase Price.
- (b) Within ninety (90) days following the Closing Date (the "Closing Statement Delivery Deadline"), Buyer shall prepare and deliver to Seller a statement (the "Closing Statement") setting forth Buyer's calculation of the Purchase Price and the related components thereof, as of the relevant calculation times. The Closing Statement shall be prepared in accordance with the Accounting Principles and the applicable definitions set forth in this Agreement.
- (c) During the thirty (30) days immediately following receipt by Seller of the Closing Statement, and any period of dispute with respect thereto thereafter, Buyer shall, and shall cause the Company to, (i) provide Seller with reasonable access during normal business hours and upon reasonable advance written notice to Buyer to the books and records of the Company for purposes of the review of the Closing Statement, and (ii) reasonably cooperate with Seller in connection with such review, provided that such access shall not unreasonably disrupt the normal operations of Buyer or its Affiliates. The Closing

Statement and the calculation of the Purchase Price set forth therein shall become final and binding upon the parties hereto thirty (30) days following receipt by Seller thereof unless Seller gives written notice of its disagreement (a "Notice of Disagreement") to Buyer prior to the end of such thirty (30)-day period; provided that the Closing Statement and the calculation of the Purchase Price set forth therein shall become final and binding upon the parties hereto upon delivery by Seller, prior to the expiration of such thirty (30)-day period, of written notice to Buyer of its acceptance of the Closing Statement and the calculation of the Purchase Price set forth therein. Any Notice of Disagreement shall specify in reasonable detail the nature and amount of any disagreement so asserted and include all supporting schedules, analyses, working papers and other relevant documentation.

(d) If a timely Notice of Disagreement is delivered by Seller, then the Closing Statement (as revised in accordance with this Section 2.3(d)), and the calculation of the Purchase Price set forth therein, shall become final and binding upon the parties hereto on the earlier of (i) the date any and all matters specified in the Notice of Disagreement are finally resolved in writing by Seller and Buyer and (ii) the date any and all matters specified in the Notice of Disagreement not resolved by Seller and Buyer are finally resolved in writing by the Independent Accountant. The Closing Statement shall be revised to the extent necessary to reflect any resolution by Seller and Buyer and any final resolution made by the Independent Accountant in accordance with this Section 2.3(d). During the thirty (30) days immediately following the delivery of a Notice of Disagreement or such longer period as Seller and Buyer may agree in writing, Seller and Buyer shall seek in good faith to resolve in writing any differences which they may have with respect to any matter specified in the Notice of Disagreement. During such period, Buyer shall be permitted to review the working papers of Seller and its Representatives and advisors relating to the Notice of Disagreement. At the end of such thirty (30)-day period or such longer period as Seller and Buyer may agree to in writing, Seller and Buyer shall submit to the Independent Accountant for review and resolution of any and all matters (but only such matters) which remain in dispute and which were included in the Notice of Disagreement. Buyer and Seller shall instruct the Independent Accountant to, and the Independent Accountant shall, make a final determination of the items included in the Closing Statement (to the extent such amounts are in dispute) in accordance with the terms of this Agreement. The Independent Accountant shall be requested to render a determination, acting as an expert and not as an arbiter, within sixty (60) days after referral of the matter to the Independent Accountant, which determination must be in writing and must set forth, in reasonable detail, the basis therefor and must be based solely on (A) the definitions and other applicable provisions of this Agreement, (B) a single written presentation (which presentation shall be limited to the matters which remain in dispute and which were specified in the timely delivered Notice of Disagreement) submitted by each of Seller and Buyer to the Independent Accountant within thirty (30) days after the engagement thereof (which the Independent Accountant shall forward to Buyer or Seller, as applicable) and (C) one written response submitted to the Independent Accountant by each of Buyer and Seller within fourteen (14) days after receipt of each such presentation (which the Independent Accountant shall forward to Buyer or Seller, as applicable), which such determination shall be conclusive, final and binding on each party hereto. No hearing shall be held and no discovery shall be permitted. Buyer and Seller will cooperate with the Independent Accountant during the term of its engagement and the parties agree not to engage in any ex parte communication with the Independent Accountant. Buyer and Seller shall instruct the Independent Accountant to, and the Independent Accountant shall, make a final determination of the items included in the Closing Statement (to the extent such items are in dispute) by selecting either Buyer's position in its entirety or Seller's position in its entirety. The Closing Statement and the calculation of the Purchase Price set forth therein shall become final and binding on the parties hereto on the date the Independent Accountant delivers its final resolution in writing to Buyer and Seller, and such resolution by the Independent Accountant shall not be subject to court review or otherwise appealable. The costs and expenses of the Independent Accountant shall be borne by the Seller, on the one hand, or Buyer, on the other hand, in such amount of disputed items submitted to the Independent Accountant based on the proportion that the aggregate amount of disputed items submitted to the Independent Accountant that is unsuccessfully disputed by Seller, on one hand, or Buyer, on the other hand, as determined by the

Independent Accountant, bears to the total amount of such disputed items so referred to the Independent Accountant for resolution (for example, should the items in dispute total in amount to \$1,000 and the Independent Accountant awards \$600 in favor of Seller's position, sixty percent (60%) of the costs of its review would be borne by Buyer and forty percent (40%) of the costs would be borne by Seller). The proportions of such fees and disbursements shall be calculated by the Independent Accountant and shall be reflected in its written report.

- (e) If the Purchase Price as finally determined in accordance with this <u>Section 2.3</u> (the "Final Purchase Price") is greater than or equal to the Estimated Purchase Price, then, within seven (7) Business Days after the Closing Statement becomes final and binding in accordance with the terms hereof (such date the Closing Statement becomes final and binding, the "Finalization Date"), (i) Buyer and Seller shall execute and deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release the Adjustment Escrow Amount to the Seller; and (ii) Buyer shall make or cause to be made payment to Seller by wire transfer of immediately available funds to the account designated by Seller, of an amount (which may be zero) equal to the amount by which the Final Purchase Price exceeds the Estimated Purchase Price.
- (f) If the Final Purchase Price is less than the Estimated Purchase Price, then, within seven (7) Business Days of the Finalization Date, (i) Buyer and Seller shall execute and deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release and pay to Buyer from the Adjustment Escrow Amount such deficit, and (ii) distribute to Seller, by wire transfer of immediately available funds to an account specified by Seller, the Adjustment Escrow Amount (after subtracting such deficit). In the event that the Adjustment Escrow Amount is not sufficient to satisfy such deficit, then Seller and Seller Member shall, within seven (7) Business Days of the Finalization Date, jointly and severally, pay to Buyer, by wire transfer of immediately available funds to an account specified by Buyer, the amount of the deficit not satisfied from the Adjustment Escrow Amount.
- (g) Payments made pursuant to this <u>Section 2.3</u> shall be treated as an adjustment to the Purchase Price for federal, state, local and non-U.S. income tax purposes, except as otherwise required by applicable Law.
- 2.4 <u>Withholding</u>. Notwithstanding anything in this Agreement to the contrary, Buyer or its designee (and, in the case of any compensatory amounts, the Company), any Affiliate of the foregoing and any other applicable withholding agent shall be entitled to withhold and deduct from the consideration otherwise payable pursuant to this Agreement such amounts as Buyer or its designee or the Company or such Affiliate or any other applicable withholding agent are required to deduct and withhold with respect to the making of such payment under any provision of applicable Law. All such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding were made.

ARTICLE III

REPRESENTATIONS AND WARRANTIES CONCERNING THE COMPANY, SELLER AND SELLER MEMBER

As a material inducement to Buyer to enter into this Agreement and consummate the transactions contemplated hereby, each of Seller and Seller Member, jointly and severally, hereby represent and warrant to Buyer as of the Closing that:

3.1 Corporate Organization; Power and Authority; Legal Capacity.

- (a) The Company is an entity duly created, formed or organized, validly existing and in good standing under the Laws of the jurisdiction of its creation, formation or organization and is qualified to do business in every jurisdiction in which its ownership of property or conduct of business requires it to qualify, except where the failure to be so qualified would not reasonably be expected to have a Material Adverse Effect. The Company possesses all requisite power and authority necessary to own and operate its properties, to carry on its businesses, to execute and deliver this Agreement and the other Transaction Documents to which it is a party and to carry out the transactions contemplated hereby and thereby. Seller has made available to Buyer true, correct and complete copies of the Company's Governing Documents which reflect all amendments made thereto. The Company is not in default under or in violation of any provision of its Governing Documents. Schedule 3.1(a) sets forth a true and correct list of (i) all of the officers, managers or directors, or members of the Company and (ii) each jurisdiction in which the Company is qualified to conduct business as a foreign entity.
- (b) Seller is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its creation, formation or organization and is qualified to do business in every jurisdiction in which its ownership of property or conduct of business requires it to qualify, except were the failure to be so qualified could not reasonably be expected to have a Material Adverse Effect. Seller possesses all requisite power and authority necessary to own and operate its properties, to carry on its businesses, to execute and deliver this Agreement and the other Transaction Documents to which it is a party and to carry out the transactions contemplated hereby and thereby.
- (c) Seller Member is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its creation, formation or organization and is qualified to do business in every jurisdiction in which its ownership of property or conduct of business requires it to qualify, except were the failure to be so qualified could not reasonably be expected to have a Material Adverse Effect. Seller Member possesses all requisite power and authority necessary to own and operate its properties, to carry on its businesses, to execute and deliver this Agreement and the other Transaction Documents to which it is a party and to carry out the transactions contemplated hereby and thereby.

3.2 Capitalization.

The authorized, issued and outstanding Equity Interests of the Company are set forth on Schedule 3.2(a)(i). All of the issued and outstanding Equity Interests of the Company are held beneficially and of record by Seller, free and clear of all Encumbrances. At the Closing, Seller shall sell to Buyer good and valid title to the Purchased Equity, free and clear of all Encumbrances. Except as set forth on Schedule 3.2(a)(ii), the Company has no outstanding securities (including options, warrants or similar rights) convertible or exchangeable for any of its Equity Interests or containing any profit participation features, nor any rights or options to subscribe for or to purchase its Equity Interests or any stock or securities convertible into or exchangeable for its Equity Interests or any stock appreciation rights or phantom stock plan or similar equity-like arrangements. The Company is not subject to any option or obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any of its Equity Interests or any warrants, options or other rights to acquire any of its Equity Interests. The Company has not violated any foreign, federal or state securities Laws in connection with the offer, sale or issuance of its Equity Interests. All of the issued and outstanding Equity Interests of the Company have been duly authorized and validly issued and are fully paid and nonassessable. There are no agreements with respect to the voting or transfer of the Company's Equity Interests or with respect to any other aspect the Company's affairs. There are no bonds, debentures, notes or other indebtedness of the Company outstanding having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which any equityholders of the Company may vote. No event has occurred, and to the Knowledge of Seller no circumstance or condition exists, that has resulted in, or that will or would reasonably be expected to result in, any liability of the Company to any current, former or alleged holder of equity interests of the Company

in such Person's capacity (or alleged capacity) as a holder of such equity interests, whether related to the Transaction or otherwise.

- (b) All Indebtedness of the Company as of immediately prior to the Closing (including the outstanding amount thereof) is set forth on Schedule 3.2(b).
 - (c) Seller Member owns all of the outstanding Equity Interests of Seller.

3.3 Authorization; Noncontravention.

- The execution, delivery and performance of this Agreement and the other Transaction Documents to which the Company, Seller or Seller Member are a party have been duly authorized by the Company, Seller and Seller Member, as applicable, and no other act (corporate or otherwise) or other proceeding on the part of the Company, Seller or Seller Member, is necessary to authorize the execution, delivery or performance of this Agreement and the other Transaction Documents to which such Person is a party and the consummation of the transactions contemplated hereby or thereby. This Agreement has been duly executed and delivered by the Company, Seller and Seller Member and constitutes a valid and binding obligation of each of them enforceable in accordance with its terms, except as enforceability may be affected by bankruptcy, insolvency, fraudulent conveyance and other similar Laws relating to or affecting creditors' rights generally, and general equitable principles (whether considered in a proceeding in equity or at law) (such limitations, the "Enforceability Limitations"), and each of the other Transaction Documents to which the Company, Seller or Seller Member is a party, when executed and delivered by the Company, Seller or Seller Member, as applicable, in accordance with the terms hereof and thereof, shall each constitute a valid and binding obligation of such Person, enforceable in accordance with its respective terms, except as may be affected by the Enforceability Limitations. Seller has full power and authority to convey good and marketable title to all of the Purchased Equity, and upon the consummation of the transactions contemplated hereby, Buyer will receive good and marketable title to such Purchased Equity, free and clear of all Encumbrances.
- (b) Except as set forth on Schedule 3.3(b), the execution and delivery by the Company, Seller and Seller Member of this Agreement and all of the other Transaction Documents to which each of them is a party and the fulfillment of and compliance with the respective terms hereof and thereof by the Company, Seller or Seller Member do not (i) conflict with or result in a breach of the terms, conditions or provisions of, (ii) constitute a default under (whether with or without the passage of time, the giving of notice or both), (iii) result in the creation of any Encumbrance upon the Purchased Equity or any assets of any the Company pursuant to, (iv) give any third party the right to modify, terminate or accelerate any obligation under, (v) result in a violation of, or (vi) require any authorization, consent, approval, exemption or other action of or by or notice or declaration to, or filing with, any third party or any Governmental Authority, the Company's Governing Documents, or any Law to which the Company, Seller or Seller Member is subject, or any Material Contract, Permit, order, judgment or decree to which the Company, Seller or Seller Member is subject.
- 3.4 <u>Subsidiaries</u>. The Company does not have and has never had any Subsidiaries. Except as set forth on <u>Schedule 3.4</u>, the Company does not own or hold, directly or indirectly, the right to acquire any Equity Interests in any other Person or has any obligation to make any Investment in any Person.

3.5 Financial Statements.

(a) Attached to <u>Schedule 3.5(a)</u> are true and correct copies of (i) the unaudited balance sheets of the Company as December 31, 2023 and December 31, 2024 and the related statements of income for the fiscal years then ended, (ii) the unaudited balance sheet of the Company as of October 31, 2025, and

the related statement of income for the 10-month period then ended (the financial statements set forth in the foregoing clauses (i) and (ii) are collectively referred to herein as the "Financial Statements"). The unaudited balance sheet of the Company as of October 31, 2025 is referred to herein as the "Latest Balance Sheet" and such date is referred to herein as the "Latest Balance Sheet Date". Each of the Financial Statements fairly presents, in all material respects, the financial position of the Company as of the respective dates thereof and the operating results of the Company for the periods covered thereby, and has been prepared in accordance with GAAP consistently applied throughout the periods covered thereby, subject, in each case, to the absence of footnote disclosures.

- (b) All accounts receivable of the Company are (i) recorded as unconditional rights to consideration arising from sales or services in accordance with ASC 606, and (ii) measured at net realizable value, subject to an allowance for doubtful accounts as required by GAAP.
- (c) Operating expenses of the Company reflected on the Financial Statements are recognized as incurred in accordance with GAAP, including both cash and accrual basis items, and are presented without regard to timing of cash payments.
- (d) All current liabilities of the Company reflected on the Financial Statements (including credit card payables and accrued payroll), and all current liabilities arising from or otherwise relating to the Company as of the Closing Date, (i) relate to the Company, and (ii) arose in transactions in the ordinary course of business. Current liabilities are recorded in accordance with GAAP when the obligation is incurred, reflecting all material obligations and estimates.
- (e) The Company maintains a system of internal controls that provides reasonable assurance that transactions are recorded as necessary to permit the preparation of the Financial Statements in accordance with GAAP consistently applied. To the Knowledge of Seller, no fraud has been discovered, whether or not material, that (i) involved any member of the management of the Company or its Affiliates that fulfill a similar role with respect to the Company, (ii) involved any other senior employee of the Company or its Affiliates that fulfill a similar role with respect to the Company, or (iii) involved any other employee, of any position or role, that highlights any deficiencies or weaknesses in the internal controls over financial reporting of the Company.
- 3.6 Absence of Undisclosed Liabilities. Except as set forth on Schedule 3.6, the Company has not, and the Company will not have, any obligation or liability (whether accrued, absolute, contingent, unliquidated or otherwise, whether known or unknown, whether due or to become due and regardless of when or by whom asserted) arising out of any transaction entered at or prior to the Closing, or any action or inaction at or prior to the Closing, or any state of facts existing at or prior to the Closing, other than (a) liabilities to the extent specifically reflected and adequately reserved on the Latest Balance Sheet, (b) liabilities and obligations which have arisen since the Latest Balance Sheet Date in the ordinary course of business (none of which is a liability for breach of contract, breach of warranty, tort, infringement, violation of Law, claim or lawsuit), or (c) obligations under any Material Contract or under contracts and commitments entered into in the ordinary course of business which are not required to be disclosed on Schedule 3.10 pursuant to Section 3.10 (but not liabilities for any breach of any such contract or commitment occurring on or prior to the Closing Date).
- 3.7 <u>No Material Adverse Effect</u>. Since December 31, 2024, there has occurred no fact, event or circumstance which has had, or would reasonably be expected to have, a Material Adverse Effect.
- 3.8 <u>Absence of Certain Developments</u>. Except as set forth on <u>Schedule 3.8</u>, since December 31, 2024, (i) the Company has conducted its business in the ordinary course of business, and (ii) the Company has not:

- (a) amended its Governing Documents;
- (b) split, combined, exchanged or reclassified any of its Equity Interests;
- (c) issued any notes, bonds or other debt securities or any Equity Interests or any securities or rights convertible, exchangeable or exercisable into any Equity Interests;
- (d) declared or paid any dividends or distributions on or in respect of any of its Equity
 Interests or redeemed, purchased or acquired any of its Equity Interests;
 - (e) incurred, assumed or guaranteed any Indebtedness;
- (f) mortgaged or pledged any of its properties or assets (including any Intellectual Property Rights) or subjected them to any Lien, except for Permitted Liens;
- (g) acquired any material tangible or intangible assets, except in the ordinary course of business:
- (h) sold, assigned, transferred, leased, licensed (or sublicensed), or otherwise encumbered any of its material tangible or intangible assets, except in the ordinary course of business, or canceled any material debts or claims;
- (i) sold, assigned, transferred, leased, licensed (or sublicensed), granted any other rights under, encumbered, or otherwise disposed of any Intellectual Property Rights (except, in each case, for any nonexclusive licenses (or sublicenses) granted to customers or vendors in the ordinary course of business), disclosed to any Person (other than to Buyer and its Affiliates), or allowed to fall into the public domain, any material Confidential Information, or abandoned or permitted to lapse any Intellectual Property Rights;
- (j) hired, engaged, furloughed, temporary laid off or terminated (without cause) the employment or service of any current or former employee or independent contractor with a total annual compensation that exceeds \$100,000;
- (k) (i) made or granted any bonus or any compensation or salary increase or decrease to any current or former employee, consultant or other individual service provider, (ii) made or granted any increase in the benefits or compensation provided under any Plan, (iii) amended or terminated any existing Plan or other benefit or compensation plan, program, policy, agreement or arrangement, (iv) established or adopted any new Plan or other benefit or compensation plan, program, policy, agreement or arrangement or (v) entered into, negotiated, extended, amended or terminated any collective bargaining agreement or other labor agreement with any labor union or other labor organization or employment or consulting agreement;
- (l) (i) modified, amended, extended, terminated or entered into any Labor Agreement or (ii) recognized or certified any labor union, labor organization, works council, or group of employees as the bargaining representative for any employees of the Company;
- (m) implemented or announced any facility or plant closings, layoff of employees, or other actions that that could implicate the WARN Act;

- (n) waived or released any noncompetition, nonsolicitation, nondisclosure, noninterference, nondisparagement, or other restrictive covenant obligation of any current or former employee or independent contractor;
- (o) suffered any extraordinary losses or waived any rights of material value (regardless of whether in the ordinary course of business) in excess of \$50,000 in the aggregate;
- (p) delayed or postponed the payment of any accounts payable or commissions or any other liability or agreed or negotiated with any party to extend the payment date of any accounts payable or commissions or any other liability or accelerated the collection of (or discounted) any accounts or notes receivable, in each case other than in the ordinary course of business;
 - (q) made commitments for capital expenditures that aggregate in excess of \$50,000;
- (r) made any loans or advances to, guaranties for the benefit of, or any Investments in, any Person;
- (s) suffered any damage, destruction or casualty loss exceeding in the aggregate \$50,000, whether or not covered by insurance;
- (t) made any change in any method of accounting or accounting policies or made any write-down in the value of its Inventory that is material or that is other than in the usual, regular and ordinary course of business or reversed any accruals (regardless of whether in the ordinary course of business);
- (u) made or changed any Tax election, adopted or changed any tax accounting method, changed any annual Tax accounting period, filed any amended Tax Return, entered into any Tax closing agreement, settled any Tax claim or assessment, surrendered any right to claim a refund of Taxes, consented to any extension or waiver of the limitations period applicable to any Tax claim or assessment, or failed to pay any Tax that became due and payable (including estimated Tax payments);
- (v) acquired, by merger or consolidation with, or by purchase of all or a substantial portion of the assets or equity of, or by any other manner, any business or any Person or any division thereof, or entered into any joint venture, partnership or other similar arrangement for the conduct of its business, or making of any Investment in, or any loan to, any other Person;
- (w) adopted any plan of merger, consolidation, reorganization, liquidation or dissolution or filed a petition in bankruptcy under any provisions of federal or state bankruptcy Law or consented to the filing of any bankruptcy petition against it under any similar Law;
- (x) entered into any agreement or arrangement prohibiting or restricting it from freely engaging in any business or otherwise restricting the conduct of its business anywhere in the world;
- (y) made any purchase of Inventory outside the ordinary course of business or changed the Company's Inventory purchasing practices; or
 - (z) agreed, whether orally or in writing, to do any of the foregoing.

3.9 Assets.

(a) The Company has good and marketable title to, or a valid leasehold interest in or valid and enforceable license to use, all properties and assets useful or used by the Company or shown on

the Latest Balance Sheet or acquired after the date thereof (other than properties and assets disposed of for fair consideration in the ordinary course of business since the Latest Balance Sheet Date), free and clear of all Liens (except for Permitted Liens). The Company owns, has a valid leasehold interest in or has the valid and enforceable right to use, all assets, properties, rights (including contractual rights), titles or interests, tangible or intangible, necessary for the conduct of its businesses as presently conducted.

- (b) All fixtures, vehicles, machinery, equipment and other tangible assets (whether owned or leased) used by the Company are, except for ordinary wear and tear and repairs being carried out in the ordinary course of business, in good condition and repair and are fit for use in the ordinary course of business of the Company as presently conducted.
- The Company does not own, and has never owned, any real property. The Company has a valid leasehold interest in each Leased Real Property. Schedule 3.9(c) sets forth a complete list of the addresses of all real property leased or subleased by the Company, the identity of the lessee or sublessee thereof and a true and complete list of all leases, subleases, licenses and rights to occupy land, buildings, structures, improvements, fixtures or other interests in real property held by the Company (the "Leased Real Property"). Seller has made available to Buyer complete and accurate copies of each of the leases for the Leased Real Property, together with all amendments, extensions, guarantees, subordination non-disturbance and attornment agreements and any other binding supplements, waivers or other changes thereto, either written or oral, under which the Company is the lessee, or holds or operates, any real property (each a "Lease" and collectively, the "Leases"). The Leased Real Property constitutes all the real property required to be owned or leased by the Company in connection with the operation of the Company's business as currently conducted. There are no eminent domain, condemnation or other similar proceedings pending or, to the Knowledge of Seller, threatened against the Company or otherwise affecting any portion of Leased Real Property, and the Company has not received any notice of the same. The current use of the Leased Real Property does not violate in any material respect any instrument of record or agreement affecting the Leased Real Property, and there is no violation of any covenant, condition, restriction, easement or order of any Governmental Authority having jurisdiction over the Leased Real Property or the use or occupancy thereof, except for such violations as would not materially interfere with the continued use and operations for the Company's business as currently conducted on the property to which they relate or materially adversely affect the value thereof for the current use of the Company. With respect to each Lease: (i) such Lease is legal, valid, binding and enforceable against the Company and, to the Knowledge of Seller, each other party thereto, and in full force and effect; (ii) neither the Company nor, to the Knowledge of Seller, any other party to such Lease is in breach or default thereof, nor does there exist any condition that, upon the passage of time or the giving of notice or both, would cause a default, nor has any waiver, indulgence or postponement of any of the Company's obligations, as party to such Lease, been granted by any owner of Leased Real Property; (iii) the Company has not assigned, transferred, conveyed, mortgaged, deeded in trust, encumbered or subleased any interest in such Lease; (iv) the possession and quiet enjoyment of the Leased Real Property under such Lease by the Company has not been disturbed and there are no disputes with respect to such Lease; (v) no security deposit or portion thereof deposited with respect to such Lease has been applied in respect of a breach or default under such Lease which has not been redeposited in full; (vi) the Company does not owe, and there is no agreement or obligation existing as of the Closing regarding the payment by the Company of any future, any brokerage commissions or finder's fees with respect to such Lease; (vii) the Company has not subleased, licensed, or otherwise granted any person the right to use or occupy, all or any portion of the Leased Real Property subject to such Lease; (viii) leasehold interests relating to such Lease are free and clear of all Liens, other than Permitted Liens; (ix) the Company has not received any written notice from the other party to any Lease of the termination or proposed termination thereof; (x) no Person other than the Company has the right to use, occupy or lease all or any portion of the Leased Real Property; and (xi) the Company has not received any notice of any violation or claimed violation by any of them of any such Laws with respect to the Leased Real Property which have not been resolved.

3.10 Contracts and Commitments.

- (a) Except as set forth on <u>Schedule 3.10</u>, the Company is not party to or bound by any written or oral:
- (i) pension, profit sharing, stock option, stock appreciation right, phantom stock, employee stock purchase, equity-based incentive or other plan or arrangement providing for deferred or other compensation (including any bonuses or other remuneration and whether in cash or otherwise), to current or former employees, consultants or other individual service providers, or any other employee benefit plan or arrangement, or any severance, change in control, or retention agreement, program, policy or arrangement;
- (ii) contract for the employment, service, retention or engagement of any officer, director, manager, individual employee or other Person on a full-time, part-time, individual consulting, or individual independent contractor or other basis providing for annual compensation or fees that exceed \$100,000 per year;
- (iii) settlement, conciliation or similar agreement with, or enforceable by, any Governmental Authority or pursuant to which the Company will have any material obligation or is obligated to pay consideration after the date of this Agreement;
- (iv) partnership, joint venture, or other similar contract or arrangement, or any contract relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets, or otherwise);
- (v) collective bargaining agreement or other contract with any labor union, trade union, works council, or other labor organization, or other Person representing or seeking to represent any employee of the Company or other individual who provides services to the Company (each a "Labor Agreement");
- (vi) contract under which the Company has advanced or loaned any amount to any other Person;
- (vii) agreement or indenture relating to borrowed money or other Indebtedness
 or the mortgaging, pledging or otherwise placing a Lien on any material asset or group of assets of the
 Company;
 - (viii) Guaranty;
 - (ix) contract for sales and marketing services;
- (x) any contract, settlement or other similar agreement with Governmental Authorities;
- (xi) lease or agreement under which the Company is lessee of or holds or operates any property, real or personal, owned by any other party, except for any lease of personal property under which the aggregate annual rental payments do not exceed \$50,000;
- (xii) lease or agreement under which the Company is lessor of or permits any third party to hold or operate any property, real or personal, owned or controlled by the Company;

- (xiii) contract or group of related contracts with the same party or group of affiliated parties the performance of which by any party thereto involves aggregate consideration in excess of \$100,000 during any twelve (12)-month period, other than purchase and sales orders entered into in the ordinary course of business;
- (xiv) contract or agreement (A) relating to the licensing (or sublicensing) of any Intellectual Property Right by the Company to a third party or by a third party to the Company (except licenses to the Company of commercially available, unmodified, "off the shelf" Software used solely for the Company's own internal use for an aggregate fee, royalty or other consideration for any such Software or group of related Software licenses of no more than \$10,000), (B) under which it is a licensor or otherwise grants to a third party any rights to use any Intellectual Property Right (other than Intellectual Property Rights licensed to customers on a non-exclusive basis in the ordinary course of business), (C) for the development of Intellectual Property Rights for the benefit of the Company, (D) for the development of Intellectual Property Rights for the benefit of any Person, (E) affecting the Company's ability to use or enforce any Intellectual Property Right (including concurrent use agreements, settlement agreements, and covenant not to sue agreements), or (F) for the licensing, ingestion, transfer or sale of Business Data;
- (xv) warranty agreement (or agreement containing warranty provisions) with respect to services rendered or products sold or leased by the Company;
 - (xvi) sales, distribution, manufacturing, supply or franchise agreement;
- (xvii) contract or agreement regarding any bond, surety or indemnification provided to or by the Company, including any contract regarding any indemnification or assumption of liability provided with respect to Environmental and Safety Requirements;
 - (xviii) contract with a Material Customer;
 - (xix) contract with a Material Vendor;
- (xx) any contract containing any provision or covenant that binds or purports to bind "Affiliates" of the Company or any of their respective assets or that would otherwise bind or purport to bind Buyer or any of its Affiliates, or any of its or their respective assets, after the Closing;
 - (xxi) any outstanding powers of attorney executed on behalf of the Company;
- (xxii) contract which provides for or otherwise includes a minimum volume or purchase requirement or similar obligation or arrangement; or
- (xxiii) contract or agreement containing any of the following provisions enforceable against the Company: (A) most favored customer pricing covenants; (B) non-competition with any Person in any business; (C) prohibitions on engagement in any business in any market or geographic area or during any time period; (D) non-solicitation of clients or customers; (E) non-solicitation and/or non-hire of any individual; (F) grants of exclusive rights, rights of first refusal, rights of first negotiation, or similar rights to any Person or (G) any other contract or agreement prohibiting the Company from freely engaging in any business or competing anywhere in the world.
- (b) All of the Material Contracts are binding and enforceable in accordance with their respective terms, and will be in full force and effect without penalty in accordance with their respective terms upon consummation of the transactions contemplated hereby. The Company has performed all

obligations required to be performed by the Company and is not in default under or in breach of nor in receipt of any claim of default or breach under any Material Contract. No event has occurred which with the passage of time or the giving of notice or both would result in a default, breach or event of noncompliance by the Company under any Material Contract. The Company does not have any present expectation or intention of not fully performing all of its obligations under the Material Contracts. To the Knowledge of Seller, no other party to any Material Contract is in breach thereof or has any present expectation or intention of not fully performing all of its obligations under such Material Contract.

- (c) The Company has not received any written or, to the Knowledge of Seller, oral notice from any other party to any Material Contract to the effect that any such party will and, to the Knowledge of Seller, no such party will, cancel, terminate or materially change the business relationship between such party and the Company (including with respect to the supplying or purchasing of materials, products or services to or from the Company, as applicable) whether as a result of the consummation of the transactions contemplated hereby or otherwise. The Company is not, and the Company has not been, involved in any claim, dispute or controversy with any other party to any Material Contract.
- (d) Seller has made available to Buyer a true and correct copy of each written Material Contract and an accurate description of each oral Material Contract, together with all amendments, waivers, modifications or other changes thereto.

3.11 Intellectual Property Rights.

- (a) Schedule 3.11(a) contains a complete and accurate description and list of (i) all registered Intellectual Property Rights owned, or purported to be owned, or exclusively licensed by or on behalf of, the Company (collectively, the "Registered Company Intellectual Property") and (ii) any unregistered Intellectual Property Rights owned, or purported to be owned, or exclusively held by the Company that are material to the conduct of the Company's business as presently conducted and as presently proposed to be conducted. Each item of Registered Company Intellectual Property is valid, subsisting, and enforceable, all necessary registration, maintenance and renewal fees currently due in connection with such Registered Company Intellectual Property have been filed with the relevant patent, copyright, trademark or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Registered Company Intellectual Property. The Company has not misrepresented, or failed to disclose, any facts or circumstances in any application for any Registered Company Intellectual Property that would constitute fraud with respect to such application.
- (b) Except as set forth on Schedule 3.11(b) and other than with respect to those Intellectual Property Rights to be provided pursuant to the Transition Services Agreement, the Company (i) exclusively owns and possesses all right, title and interest in and to the Intellectual Property Rights, including that set forth (or required to be set forth) on Schedule 3.11(a), and (ii) exclusively owns and possesses all right, title and interest in and to, or have the right to use pursuant to a valid and enforceable written license, all other Intellectual Property Rights necessary for or used in the operation of its business as presently conducted and as presently proposed to be conducted (the Intellectual Property Rights in the foregoing clause (i) and this clause (ii), collectively, the "Company Intellectual Property"), in each case for clause (i) and (ii) of this Section 3.11(b), free and clear of all Liens except for Permitted Liens. The Company Intellectual Property shall be available for use by Buyer and the Company immediately after the Closing Date on identical terms and conditions to those under which the Company owned or used the Company Intellectual Property immediately prior to the Closing Date.

- (c) Except as set forth on Schedule 3.11(c), the Company has not developed, or has not had developed on its behalf, any material Software or Company Products that is used in the Company's businesses, and the Company does not, and has not in the past, provide or offer to any Person any Company Products. The Company has not disclosed, delivered, licensed or otherwise made available, and the Company does not have a duty or obligation (whether present, contingent or otherwise) to disclose, deliver, license or otherwise make available, any source code developed or created by the Companies to any Person, other than to (i) the Person on whose behalf the Company has developed or created such source code, or (ii) the employees or independent contractors who have been employed or engaged by the Company to develop such source code on the Company's behalf. The Company has not entered into any source code escrow agreements.
- (d) The Company owns and possesses all right, title and interest in and to all Intellectual Property Rights created or developed by or on behalf of, or otherwise under the direction or supervision of, the Company's employees or independent contractors, relating to the Company's business. All Persons who have contributed to the creation, invention, modification, or improvement of any Intellectual Property Rights (i) owned, or purportedly owned by the Company, in whole or in part, or (ii) developed by the Company on behalf of any Person, in whole or in part, have signed enforceable written agreements (A) assigning to the Company all Intellectual Property Rights created or developed within the scope of employment or engagement, as applicable, and (B) protecting the trade secrets and Confidential Information of the Company, and all such assignment agreements and confidentiality agreements are, in each case, valid and are enforceable in accordance with their terms and, to the Knowledge of Seller, no Person is in breach of any such agreement.
- Except as set forth on Schedule 3.11(e), the loss or expiration of any Company (e) Intellectual Property has not had and would not reasonably be expected to be material to the Company, and no loss or expiration of any Company Intellectual Property is threatened, pending or reasonably foreseeable (other than the expiration of any issued or registered Intellectual Property Rights at the end of their respective statutory terms). The Company has taken all commercially reasonable and necessary steps to maintain and protect the Company Intellectual Property, including the secrecy, confidentiality and value of trade secrets and other confidential information of the Company and any third parties who have shared their confidential information with the Company. The Company has not disclosed any confidential Company Intellectual Property (including the source code to any Company Products) to any third party other than pursuant to a written confidentiality agreement pursuant to which such third party agrees to protect such confidential information. Each of the Intellectual Property Rights (or applications therefor) set forth or required to be set forth on Schedule 3.11(a) is subsisting and in full force and effect, is valid and, to the Knowledge of Seller, is enforceable. All renewal and maintenance fees in respect of each item of Intellectual Property Rights (or applications therefor) set forth on Schedule 3.11(a) have been duly paid and none of the registrations or applications are subject to any challenge, opposition, nullity proceeding or interference or threats to commence the same.
- (f) There is no unauthorized use, unauthorized disclosure, infringement, dilution or misappropriation by any third party of any Company Intellectual Property that is owned, or purported to be owned, by the Company (collectively, the "Company Owned Intellectual Property"). The Company has not brought any Action for infringement, dilution or misappropriation of any Company Owned Intellectual Property. No Company Owned Intellectual Property nor, to the Knowledge of Seller, any other Company Intellectual Property, is infringing, diluting, misappropriating or violating, and has not infringed, misappropriated or violated, the Intellectual Property Rights of any third party. The Company has not been sued in any Action, nor received any written communications alleging that any Company Intellectual Property, or that the Company has violated or, by conducting its businesses, would violate any Intellectual Property Rights of any third party.

- (g) The Company does not use and has not used any Open Source Software or any modification or derivative thereof (i) in a manner that would grant or purport to grant to any Person any rights to or immunities under any of the Company Owned Intellectual Property, or (ii) under any license requiring the Company to disclose or distribute the source code to any of the Company Products or Software that is developed by or on behalf of the Company and used in the Company's business, to license or provide the source code to any of the Company Products for the purpose of making derivative works, or to make available for redistribution to any Person the source code to any of the Company Products at no or minimal charge.
- (h) The Company has been and is in material compliance with all obligations under any agreement pursuant to which the Company has obtained the right to use any third party Software, including Open Source Software, and in particular the Company has purchased a sufficient number of seat licenses for the Systems.
- (i) Except as set forth on Schedule 3.11(i), the Company owns, leases, licenses, or otherwise has the legal right to use all Systems. All Systems owned or used by or for the Company (i) are free from any Malicious Code or other programming, design or documentation error or corruption or other software routines or hardware components that permit unauthorized access or the unauthorized disablement or erasure of such, (ii) are fully functional and operate and run in a reasonable and efficient business manner and (iii) are sufficient for the current and currently contemplated needs of the business of the Company including as to capacity, scalability, and ability to meet current peak volumes and anticipated volumes in a timely manner, and there have been no failures, breakdowns, outages, or unavailability of any of the foregoing Software or Systems in the twenty-four (24) months prior to the Closing Date. The Company has taken all commercially reasonable steps to safeguard the internal and external integrity and security of the Software and Systems owned or used by or for the Company, including procedures aimed at preventing unauthorized access and the introduction of Malicious Code. The Company has implemented and maintains commercially reasonable business continuity and disaster recovery plans, procedures, and facilities and all such plans have been proven effective upon regular testing in all material respects.
- (j) There is no Malicious Code in any of the Company Products or any of the Software that is developed by or on behalf of the Company and used in the Company's business, and the Company has not received any complaints from any Person related to any Malicious Code.
- The Company has commercially reasonable physical, technical, organizational and administrative safeguards in place to protect Personal Information or other confidential or proprietary information of any Person (including Business Data) in its possession or control from unauthorized access or unauthorized Processing by any Person. With respect to the Processing of Company Data and Personal Information, the Company and the conduct of its businesses, has complied with, and is in compliance with all Data Security Requirements, as well as any obligations under agreements with third parties, respectively. The Company has a valid and legal right (whether contractually, by law or otherwise) to access or use all Personal Information and any other information of any Person that is accessed and used by or on behalf of the Company. The Company has not experienced any Security Incident, phishing incident, ransomware or malware attack, malicious disruption of Systems or other incident in which Personal Information or Business Data in the possession or control of the Company, was or may have been accessed, disclosed, acquired, or exfiltrated in an unauthorized manner, and the Company has not received any notices from any Personal or has been subject to any actual or threatened claim, processing or investigation with respect thereto, and, to the Knowledge of Seller, any of the Company's contractors was or may have been stolen or improperly accessed, deleted or corrupted, including any breach of security or any notices or complaints from any Person regarding any such information. The transactions contemplated by this Agreement will not result in any liabilities in connection with any Data Security Requirements. The Company has not received any written or other complaint, claim, demand, inquiry, or other notice, including a notice of

investigation, from any Person (including any governmental, regulatory or self-regulatory authority or entity) regarding the Processing of Personal Information or alleging that the Company's Processing of Personal Information is in violation of any applicable Data Security Requirement or otherwise constitutes an unfair, deceptive, or misleading trade practice. The Company has obtained written agreements from all subcontractors and third-party vendors to whom the Company has provided or disclosed Personal Information that (1) satisfy the requirements of the Data Security Requirements, and (2) bind the subcontractor and third-party vendors to at least the same restrictions and conditions that apply to the Company with respect to such Personal Information he Company has taken commercially reasonable steps to limit access to Personal Information to: (i) the Company personnel and to subcontractors and third-party vendors providing services to or on behalf of the Company, in each case to those who have a need to know such Personal Information in the execution of their duties to the Company; and (ii) such other Persons permitted to access such Personal Information in accordance with the Privacy Policies, and contractual obligations to which the Company is bound.

- (l) Schedule 3.11(l) sets forth a true, correct and complete list of all: (i) third-party AI Tools (together with: (A) the license terms applicable to each AI Tool, (B) the purposes for which the Company has used each AI Tool, and (C) a general description of the material outputs generated by each AI Tool and the uses made by the Company of such outputs), (ii) Company AI Products and (iii) Training Datasets. The Company has obtained and complied with all licenses, consents, agreements, terms, conditions and permissions applicable to each Training Dataset. The Company has implemented and maintained appropriate controls, policies, procedures, safeguards, measures, plans and technologies with respect to the Company's use of Generative AI as appropriate based on the Company's use thereof to mitigate risks of infringement, misappropriation or other violations of the rights of other Persons and protection of Confidential Information.
- Except as set forth on Schedule 3.11(m), (i) there are no, and there have been no (m) claims made against the Company, Seller or Seller Member with respect to the validity, infringement, use, ownership, or enforceability of any Company Intellectual Property and, to the Knowledge of Seller, there is no basis for any such claim, (ii) none of the Company, Seller or Seller Member has received any notice of, and to the Knowledge of Seller, there are no facts which indicate a likelihood of, any infringement, dilution, misappropriation, or other violation by, or conflict with, any third party with respect to any Intellectual Property Rights (including any demand or request that the Company license any rights from a third party) nor has the Company requested or received any opinions of counsel related to the same, (iii) the conduct of the Company's business does not, and has not infringed, diluted, misappropriated, or otherwise violated or conflicted with, and the continued conduct of the Company's business will not infringe, dilute, misappropriate, or otherwise violate or conflict with, any Intellectual Property Rights of other Persons, (iv) no third party has made any written claim asserting that any Intellectual Property Rights owned or held by the Company should be transferred to or placed under the control of a third party, nor has any third party made a request or demand that any such transfer be made by the Company, and (v) to the Knowledge of Seller, the Intellectual Property Rights owned by or licensed to the Company have not been infringed, misappropriated, or otherwise violated or conflicted with by other Person.
- (n) Except as set forth on <u>Schedule 3.11(n)</u>, the transactions contemplated by this Agreement and the other Transaction Documents will not have an adverse effect on the Company's right, title or interest in and to the Company Intellectual Property and all Company Intellectual Property shall be owned or available for use from a third party by the Company on terms and conditions immediately after the Closing identical to those under which they were owned or available for use immediately before the Closing.
- 3.12 <u>Litigation</u>. Except as set forth on <u>Schedule 3.12</u>, there are no, and since January 1, 2021 there have not been any, Actions, suits, complaints, litigation, charges, mediations, proceedings (including

any Governmental Authority or arbitration proceedings), orders, audits, inquiries, investigations, claims or any similar proceedings (collectively, "Proceedings") pending or, to the Knowledge of Seller, threatened against or affecting the Company (or to the Knowledge of Seller, pending or threatened against or affecting any of the officers, directors, managers, direct and indirect equityholders, or employees of the Company with respect to its business activities), or pending or threatened by the Company against any Person, at Law or in equity, or before or by any Governmental Authority (including any Proceedings with respect to the transactions contemplated by this Agreement and the other Transaction Documents). The Company is not, or since January 1, 2021 has not been, subject to any judgment, order, settlement, award, injunction, ruling, writ or decree of any Governmental Authority, and the Company has not received any opinion or memorandum or advice from legal counsel to the effect that it is exposed, from a legal standpoint, to any material liabilities. There are no Proceedings pending or, to the Knowledge of Seller, threatened against or affecting Seller or the Company in which it is sought to restrain or prohibit or to obtain damages or other relief in connection with the transactions contemplated hereby.

3.13 Compliance with Laws. Except as set forth on Schedule 3.13:

- (a) The Company has, since January 1, 2021, complied, and is now in compliance, with all applicable Laws relating to the operation of its business and the maintenance and operation of its properties and assets. Since January 1, 2021, no written or, to the Knowledge of Seller, oral notice has been received by the Company alleging a violation of any such Laws and, to the Knowledge of Seller, the Company has not been subject to any inspection, finding, investigation, penalty assessment, audit or other compliance or enforcement action.
- (b) Except as set forth on Schedule 3.13, the Company has all Permits required to conduct the Company's business as currently conducted. Schedule 3.13 lists and describes all material Permits held by the Company. All such Permits, if any, (i) have been issued or given to the Company and no other Person and are in good standing and full force and effect and (ii) constitute all licenses, permits, registrations, accreditations, certifications, approvals and agreements and consents that are required for the Company to conduct its businesses as currently conducted (including the receipt of payment or reimbursement from customers). The Company is operating, and since January 1, 2021, has operated, in material compliance with each such issued Permit. To the Knowledge of Seller, no Governmental Authority has initiated any investigation, audit, or other action which would reasonably be expected to result in the suspension, revocation, limitation, or modification of any Permit held by the Company. To the extent legally permissible, all such Permits will be available for use by the Company immediately after the Closing.
- (c) The Company has never applied for or received any CARES Act loans or proceeds (including, but not limited to, a PPP Loan and Economic Injury Disaster Loans).

3.14 Environmental, Health and Safety Matters. Except as set forth on Schedule 3.14:

- (a) The Company has complied, and is now in compliance, in all material respects, with all applicable Environmental and Safety Requirements.
- (b) Without limiting the generality of the foregoing, the Company has obtained and complied with, and is in compliance with, in all material respects, all Permits and other authorizations that are required pursuant to Environmental and Safety Requirements.
- (c) The Company has not received any notice, report, order, directive, or other information regarding or relating to any actual or alleged violation of Environmental and Safety Requirements, or any liabilities (whether accrued, absolute, contingent, unliquidated or otherwise),

including any investigatory, remedial or corrective obligations, relating to the Company, its past or current facilities, or its business, in each case arising under Environmental and Safety Requirements.

- (d) The Company has not, nor, to the Knowledge of Seller, any third party has not, used, treated, stored, disposed of, arranged for or permitted the disposal of, transported, handled, manufactured, distributed, Released or exposed any Person to any Hazardous Material.
- (e) To the Knowledge of Seller, the Company has not owned or operated any property or facility which is or has been contaminated by any Hazardous Material.
- (f) The Company has not manufactured, sold, marketed, installed or distributed products or other items containing any Hazardous Materials.
- (g) There is no unresolved, pending, or threatened litigation, enforcement actions, or other legal proceedings under any Environmental and Safety Requirement against the Company, including with respect to "greenwashing" or any environmental reporting obligations of the Company.
- (h) The Company has assumed, undertaken, provided an indemnity with respect to, or otherwise become subject to, any liability of any other Person relating to Environmental and Safety Requirements or Hazardous Materials.
- (i) Seller has made available to Buyer all environmental audits, reports, assessments and other material environmental, health and safety documents relating to the Company, or its current and former properties, operations and facilities.

3.15 Employees.

- (a) Schedule 3.15(a) correctly sets forth for each current employee and (as applicable), independent contractor of the Company: their (i) first and last name (or entity name, if applicable), (ii) job title, (iii) current annual salary, hourly wage rate, or fee, as the case may be, (iv) legal employer or 1099 issuer, as the case may be, (v) work location, (vi) hire or engagement date, (vii) exempt or non-exempt status (as applicable), (viii) full-time or part-time status (as applicable), (ix) type and expiration date of any required visa or work permit, (x) accrued but unpaid vacation or other paid time off, and (xi) whether any employees are absent from active employment, including leave of absence or disability (or gave written notice of the need for a leave of absence), and anticipated date of return.
- (b) Except as set forth on <u>Schedule 3.15(b)</u>, to the Knowledge of Seller, no executive or employee of the Company nor any group of employees or independent contractors of the Company (i) have any plans to terminate or alter the terms of their employment or engagement with the Company, (ii) have received an offer to join a business that may be competitive with any business of the Company, or (iii) is a party to or bound by any confidentiality agreement, noncompetition agreement or other contract (with a Person other than the Company) that may reasonably be expected to have an adverse effect on the performance by such employee of any of his or her duties or responsibilities as an employee of the Company. All employees of the Company are employed on an "at will" basis and their employment can be terminated at any time for any reason without any amounts being owed to such individual. There are no outstanding offers of employment made by the Company to any individual.
- (c) The Company has complied at all times, and is in material compliance, with all Laws relating to employment, employment practices, labor, and retention of employees and independent contractors (including all Laws relating to terms and conditions of employment, health and safety, employment harassment, discrimination or retaliation, whistleblowing, disability rights or benefits, plant

closures and layoffs (including the WARN Act and any similar Laws), employee trainings and notices, restrictive covenants, workers' compensation, labor relations, employee leave issues, termination, collective dismissals, unemployment insurance, COVID-19, affirmative action, unemployment insurance wages and hours (including the classification of independent contractors and exempt and non-exempt employees), pay transparency, equal opportunity, immigration (including the completion, processing, and maintenance of Forms I-9 for all employees and the proper confirmation of employee visas), collective bargaining, labor relations, the payment of social security and other Taxes, and notices and automated employment decision tools and other artificial intelligence). To the Knowledge of Seller, no employee or independent contractor of the Company is in violation of any contract, including a noncompete, nondisclosure, nonsolicitation, confidentiality, employment, consulting or similar agreement or common law nondisclosure obligation, fiduciary duty, restrictive covenant, or other obligation: (i) owed to the Company; or (ii) owed to any third party with respect to such person's right to be employed or engaged by the Company. Schedule 3.15(c) sets forth the bonuses paid or payable to each officer and employee of the Company in respect of the fiscal year ended December 31, 2024 and the maximum bonus which may be earned by each such officer or employee in respect of the fiscal year ending December 31, 2025.

- (d) The Company has never been, a party to, or bound by, any Labor Agreement or other labor relationship with any labor union, works council or other labor organization, nor is the Company obligated to recognize or bargain with any labor organization or group of employees as the representative of any of its employees. No employees of the Company are represented by any labor union, works council, other labor organization or other employee representative or group of employees with respect to their employment with the Company. To the Knowledge of Seller, there are no union organizational or decertification activities underway or threatened by, on behalf of or against any labor organization or group of employees with respect to any employees of the Company, and no such activities have occurred. Since January 1, 2021, there have been no pending, or to the Knowledge of Seller, threatened, strikes, slowdowns, work stoppages, lockouts, unfair labor practice complaints or charges, picketing, handbilling, labor grievances, labor arbitrations or other labor disputes against or affecting the Company, no such disputes are pending or, to the Knowledge of Seller, threatened, and there are no facts or circumstances that could reasonably be expected to give rise to any strikes, slowdowns, work stoppages, lockouts, unfair labor practice complaints or charges, picketing, handbilling, labor grievances, labor arbitrations or other labor disputes. With respect to the transactions contemplated by this Agreement, any notice, consultation, consent or similar obligations owed to employees or their representatives required under any Law, contract or Labor Agreement has been given, and all bargaining obligations with any employee or employee representative have been satisfied.
- (e) The Company has not (i) implemented any layoff of employees or facility or plant closings that could trigger the WARN Act or any similar Laws, (ii) issued any notification of a plant closing or mass layoff required by the WARN Act or any similar Laws, or (iii) incurred any liability or obligation that remains unsatisfied under the WARN Act or any similar Laws.
- (f) The Company is not delinquent in payment of, and has not failed to pay, any of its current or former employees, consultants, leased employees, independent contractors or other non-employee service providers wages (including minimum wage, overtime, wage premiums, meal breaks or waiting time penalties), salaries, fees, commissions, accrued and unused vacation, on-call payments, equal pay or collective bargaining payments, bonuses, severance and termination payments or other compensation, if any, to which they would be entitled under applicable Law, company policy, or agreement. The Company does not have any accrued unpaid liabilities relating to its current and former employees, directors, independent contractors or other service providers, other than for compensation that has accrued since the last regularly scheduled payroll date in the ordinary course of business. The Company has properly classified all employees under the Fair Labor Standards Act and all other applicable Laws and there is no Action pending, threatened, or reasonably anticipated that challenges such classifications and there are no

facts that could reasonably be expected to give rise to such Action. The Company is not liable for any payment to any trust or other fund or to any Governmental Authority, with respect to unemployment compensation benefits, social security or similar benefits or obligations for employees, consultants or independent contractors (other than routine payments to be made in the normal course of business and consistent with past practice).

- (g) All Persons who perform or have performed services for the Company are either United States citizens or are legally entitled to work in the United States under the Immigration Reform and Control Act of 1986, as amended, and any other United States immigration Laws relating to the employment of non-United States citizens applicable in the state in which such Persons are or were employed. The Company is and has been, in compliance with all applicable Laws regarding immigration or employment of noncitizen workers, including with respect to E-Verify and Form 1-9, including collection and retention thereof. There are no pending, threatened, or, to the Knowledge of Seller, reasonably anticipated investigations of the Company by any branch or department of the U.S. Immigration and Customs Enforcement or other federal agency charged with administration and enforcement of federal immigration Laws.
 - (h) The Company has not engaged in any unfair labor practice.
- (i) The Company's contracts and other understandings with independent contractors have at all times complied with all applicable requirements of all Laws, rules, and regulations of any Governmental Authority. All such contracts or arrangements constitute a bona fide agreement whereby each such independent contractor is an independent contractor to, and not an employee of, the Company and has been properly classified and treated as such pursuant to the Code and all other applicable Laws, and there are no, and have been no Actions or allegations pending or, to the Knowledge of Seller, threatened at law or in equity before any Governmental Authority or by any Person that challenge (i) the Company's compliance under any rule, regulation or law, (ii) the independent contractor nature of such contracts or independent contractor's work status, or (iii) other understandings or arrangements of any nature whatsoever and there are no facts or circumstances that could reasonably be expected to give rise to any such Actions.
- (j) The Company is and has at all times been in material compliance with all Laws relating to occupational health and safety, including the Occupational Safety and Health Act of 1970, as amended, and the rules and regulations promulgated thereunder.
- (k) There have not been (i) any allegations or formal or informal complaints made to or filed with the Company related to sexual harassment or sexual misconduct or other unlawful harassment or discrimination; (ii) any other claims or Actions initiated, filed or, to the Knowledge of Seller, threatened, against the Company related to sexual harassment or sexual misconduct or other unlawful harassment or discrimination, and there are no facts or circumstances that could reasonably be expected to give rise to any such claims or Actions; and (iii) any other allegations, formal or informal complaints or any other claims initiated, filed or, to the Knowledge of Seller, threatened against any Person related to sexual harassment or sexual misconduct or other unlawful harassment or discrimination, in each case by or against any current or former employee, director, independent contractor, or other service provider of the Company.

3.16 Employee Benefit Plans.

(a) The attached <u>Schedule 3.16(a)</u> sets forth an accurate and complete list of each Plan. For purposes of this Agreement, "Plan" means each "employee benefit plan" (as such term is defined in Section 3(3) of ERISA, whether or not subject to ERISA) and each other bonus, commission, deferred compensation, retention, employment (including offer letters), consulting, change in control, sale or

transaction incentive, stock purchase, restricted stock, restricted stock unit, stock appreciation right, phantom equity, performance awards, equity option or equity-based incentive, retirement, pension, profit sharing, severance, termination pay, sick leave, vacation, paid time off, health, welfare, post-employment welfare, supplemental income arrangements, employee loans, disability insurance, life insurance, fringe benefit or other benefit or compensation plan, program, policy, agreement, understanding or arrangement, in each case, whether written or unwritten, funded or unfunded, (i) that is maintained, sponsored, contributed to or required to be contributed to by the Company or any ERISA Affiliate for the benefit of current or former employees, officers, directors or individual consultants of the Company (or their spouses, dependents or beneficiaries), or (ii) under or with respect to which the Company or any ERISA Affiliate has any current or contingent liability or obligation. The Company does not have a contract, plan or commitment, whether legally binding or not, to create any additional Plan or to modify any existing Plan. Except as set forth on Schedule 3.16(a), the Company has reserved all rights necessary to amend or terminate each Plan that is an "employee benefit plan" within the meaning of Section 3(3) of ERISA (whether or not subject to ERISA) without the consent of any other Person or any liability (other than vested benefits in accordance with the terms of the applicable Plan). None of the Plans will be subject to any surrender fees, market value adjustment, deferred sales charges, commissions, or other fees or charges upon termination other than the normal and reasonable administrative fees associated with their amendment, transfer or termination.

- (b) No Plan is, and neither the Company nor any ERISA Affiliate sponsors, maintains, contributes to, participates in, or has ever sponsored, maintained, contributed to, or participated in, or has any current or contingent liability or obligation (including on account of an ERISA Affiliate) with respect to: (i) any "multiemployer plan" (as defined in Section 3(37) or 4001(a)(3) of ERISA); (ii) any "defined benefit plan" (as defined in Section 3(35) of ERISA), or any other plan that is or was subject to Section 412 or 430 of the Code or Section 302 or Title IV or ERISA, regardless of whether now terminated; (iii) a "multiple employer plan" (within the meaning of Section 210(a), 4063 and 4064 of ERISA or Section 413(c) of the Code); or (iv) a "multiple employer welfare arrangement" (as such term is defined in Section 3(40) of ERISA). The Company does not have any current or contingent liability by reason of at any time being treated as a single employer with any other Person under Section 414 of the Code. No Plan is a self-funded or self-insured health or welfare benefit plan.
- (c) No Plan provides, and Company has not had any obligation under any Plan or otherwise promised to provide, post-employment, post-ownership or retiree medical, health, life insurance or other welfare-type benefits to current or future retired or terminated employees or any other Person (except for limited continued medical benefit coverage required to be provided under COBRA, for which the covered individual pays the full cost of coverage). The Company and its the ERISA Affiliates have complied and are in material compliance with the requirements of COBRA as well as the Patient Protection and Affordable Care Act, including the Health Care and Education Reconciliation Act of 2010, as amended and including any guidance issued thereunder ("ACA"). The Company has not incurred (whether or not assessed), or is reasonably expected to incur or to be subject to, any Tax, penalty or other liability that may be imposed under ACA (including with respect to the reporting requirements under Sections 6055 and 6056 of the Code, as applicable) or pursuant to Sections 4980B, 4980D or 4980H of the Code.
- (d) With respect to the Plans, all required payments, premiums, contributions, reimbursements or accruals for all periods ending prior to or as of the Closing shall have been timely made (to the extent due) or properly accrued on the Latest Balance Sheet (to the extent not yet due). None of the Plans has any unfunded liabilities which are not reflected on the Latest Balance Sheet.
- (e) The Plans and all related trusts, insurance contracts and funds have been established, maintained, funded, operated, and administered in accordance with their terms and in material compliance with the applicable provisions of ERISA, the Code and other applicable Law, and nothing has

occurred and no condition exists with respect to any Plan that could result in a Tax, penalty or other liability or obligation of the Company. There has been no "prohibited transaction" within the meaning of Section 4975 of the Code or Section 406 of ERISA and no breach of fiduciary duty (as determined under ERISA) has occurred with respect to any Plan, and no reportable event, as defined in ERISA, has occurred in connection with any Plan. No audit, investigations, actions, suits or claims with respect to the Plans (other than routine claims for benefits) are pending or, to the Knowledge of Seller, threatened and there are no circumstances which could give rise to or be expected to give rise to any such audits, investigations, actions, suits or claims. All premiums, payments and/or contributions required to have been made with respect to all Plans either have been made or have been accrued in accordance with the terms of the applicable Plan and applicable Law. The Company has not filed, and the Company is not considering filing, an application under the IRS Employee Plans Compliance Resolution System or the Department of Labor's Voluntary Fiduciary Correction Program with respect to any Plan. No Plan is the subject of an examination or audit by a Governmental Authority or is engaged in self-correction or other similar program.

- (f) Each of the Plans that is intended to be qualified under Section 401(a) of the Code is so qualified and has received a currently effective favorable determination letter from the Internal Revenue Service that such plan is qualified under Section 401(a) of the Code or is entitled to rely upon a currently effective opinion letter issued to the sponsor of an IRS approved M&P or volume submitter plan document, and there are no circumstances that could reasonably be expected to adversely affect the qualified status of any such Plan.
- (g) Except as set forth on Schedule 3.16(g), Seller has made available to Buyer true and complete copies of all plan documents, related trust agreements and insurance policies for each Plan, including all amendments thereto (or a written description thereof if no written plan document exists), the most recent summary plan description and any material modifications thereto, the three (3) most recent annual reports (Form 5500 and attachments) for the Plans, nondiscrimination testing results for the three (3) most recent plan years, the most recent Internal Revenue Service determination or opinion letter for any Plan intended to be qualified under Section 401(a) of the Code, and any non-routine correspondence with any Governmental Authority related to a Plan.
- (h) Neither the execution or delivery of this Agreement nor the consummation of the transactions contemplated hereby, either alone or in combination with any other event, could, whether alone or upon the occurrence of any additional or subsequent event, entitle any current or former director, officer, employee or other individual service provider (or a beneficiary thereof) of the Company to any severance pay, unemployment compensation or any other payment or benefit under any Plan or any agreement or arrangement to which the Company is a party, accelerate the time of payment, funding or vesting of, or increase the amount or value of compensation or benefits due, to any such current or former director, officer, employee or other individual service provider (or beneficiary thereof) of the Company under any Plan or any agreement or arrangement to which the Company is a party, result in the triggering or imposition of any restrictions or limitations on the ability to terminate or amend any Plan, or result in a breach or violation of, or default under any Plan.
- (i) No Plan is subject to the laws of any jurisdiction outside the United States. No Plan provides for any tax "gross-up" or similar indemnification, reimbursement, or "make-whole" payments for the additional tax set forth under Section 409A, Section 280G or Section 4999 of the Code. No amount, economic benefit or other entitlement that could be received (including in cash or property or vesting of property) as a result of the execution, delivery and performance of this Agreement or the consummation of the transactions contemplated by this Agreement (whether alone or in conjunction with any other event, including any termination of employment or service on or following the Closing Date) by any person who could be a "disqualified individual" (as defined in Section 280G(c) of the Code) with respect to the

Company could result in any "parachute payment" as defined in Section 280G(b)(2) of the Code and the regulations thereunder or trigger the excise tax under Section 4999 of the Code.

Insurance. Schedule 3.17 contains a true and complete list of all insurance policies to which the Company is a party or which provide coverage to or for the benefit of, or with respect to, the Company or any director, manager, officer, direct or indirect equityholder, or employee of the Company in his or her capacity as such (the policies set forth or required to be set forth on Schedule 3.17, the "Insurance Policies"). Seller has delivered to Buyer true and complete copies of the Insurance Policies, as well as loss runs for the Insurance Policies (or any predecessor policies) since December 31, 2021. All premiums and fees due and owing with respect to the Insurance Policies have been timely paid in full. Schedule 3.17 also describes any self-insurance or coinsurance arrangements by or affecting the Company, including any reserves established thereunder. Except as set forth on Schedule 3.17, the Insurance Policies (and predecessor policies) do not provide for any retrospective premium adjustment or other experience-based liability on the part of the Company. The Insurance Policies (or, if applicable, any predecessor policies) are of a type and level of coverage as is required by applicable Law or any Material Contract. Except as set forth on Schedule 3.17, each Insurance Policy is in full force and effect and shall remain in full force in effect following the Closing in accordance with such policy's terms and conditions. The Company is not in default with respect to its obligations under any Insurance Policy maintained by or for the benefit of the Company. The Company has complied in all material respects with all of its obligations under each Insurance Policy. The Company has given timely notice to the insurer of all claims that may be insured thereby, and Schedule 3.17 includes a true and correct list of all claims made under the Insurance Policies (or predecessor policies) since December 31, 2021. The Company has not received written or, to the Knowledge of Seller, other notice from any insurance carrier denying or disputing any claim, the amount of any claim or the coverage of any claim made on any such insurance policy or similarly reserving rights in connection therewith.

3.18 Tax Matters.

(a) Except as set forth on Schedule 3.18(a), the Company has timely filed all income and other material Tax Returns required to be filed by it under applicable Laws and regulations, each such Tax Return has been prepared in material compliance with all applicable Law, and all such Tax Returns are complete, true and accurate in all material respects. All Taxes due and payable by the Company (regardless of whether shown or required to be shown on any Tax Return) have been paid, and the Company has withheld and paid over to the appropriate taxing authority all Taxes which they were required to withhold from amounts paid or owing to any employee, independent contractor, stockholder, creditor or other third party. The Company has timely and accurately complied with all material reporting and record keeping requirements related thereto, including filing of W-2 and 1099s (or other applicable forms).

(b) Except as set forth on Schedule 3.18(b):

- (i) the Company has not requested or been granted an extension of the time for filing any Tax Return which has not yet been filed, and no such request is currently outstanding;
- (ii) the Company has not consented to extend to a date later than the date hereof the time in which any Tax may be assessed or collected by any taxing authority or otherwise waived any statute of limitations in respect of Taxes;
- (iii) no federal, state, local, or non-U.S. Tax audits or administrative or judicial Tax proceedings are pending or being conducted with respect to the Company. The Company has not received from any federal, state, local, or non-U.S. taxing authority (including jurisdictions where the Company has not filed Tax Returns) any (A) written notice indicating an intent to open an audit or other review, (B) written request for information related to Tax matters, or (C) written notice of deficiency or

proposed adjustment for any amount of Tax proposed, asserted, or assessed by any taxing authority against the Company;

- (iv) no claim has been made in writing by a taxing authority in a jurisdiction where the Company does not file Tax Returns claiming that the Company is or may be subject to Taxes assessed by such jurisdiction;
- (v) there are no Liens for Taxes (other than for current Taxes not yet due and payable) upon the assets of the Company; and
- (vi) the Company is not party to or bound by any Tax allocation or Tax sharing agreement and has no current or potential contractual obligation to indemnify any other Person with respect to Taxes;
- (c) Schedule 3.18(c) lists all the states with respect to which the Company is required to file any corporate, income, sales and use, employee withholding, or franchise Tax Returns.
- (d) The Company has at all times correctly classified those individuals performing services as common law employees, leased employees, independent contractors or other service providers to or agents of the Company, as the case may be, and to the Knowledge of Seller, the Company has no liability as a result of any failure to properly classify such individuals (including with respect to wages, Taxes, social security, working compensation or benefit plans).
- (e) The Company (i) has never been a member of an Affiliated Group filing a consolidated federal income Tax Return and (ii) does not have any liability for the Taxes of any Person under United States Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or non-US Law), as a transferee or successor, by contract, or otherwise. Neither Buyer nor the Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) beginning after the Closing Date as a result of (A) any prepaid amount received or deferred revenue accrued by the Company on or prior to the Closing Date, (B) any change in method of accounting for a taxable period ending on or prior to the Closing Date, (C) any use of an improper method of accounting or the cash method of accounting for a taxable period ending on or prior to the Closing Date, (D) intercompany transaction or excess loss account described in Treasury Regulation under Section 1502 of the Code (or any corresponding or similar provision of state, local or non-U.S. income Tax law), (E) any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. income Tax law) executed on or prior to the Closing Date, (F) any installment sale or open transaction disposition made on or prior to the Closing Date, or (G) any election under Sections 108(i), 965 or 1400Z-2 of the Code.
- (f) The Company is not, and has never been, a party to any "reportable transaction" as defined in Section 6707A(c)(1) of the Code and Treasury Regulations Section 1.6011-4(b).
- (g) Each contract, arrangement or plan of the Company that is a "nonqualified deferred compensation plan" (as defined for purposes of Code Section 409A(d)(1)) is in documentary and operational compliance with Code Section 409A and the applicable guidance issued thereunder in all respects. The Company does not have any indemnity or gross-up obligation for any Taxes imposed under Section 4999 or 409A of the Code.
- (h) No assets of the Company constitute "Section 197(f)(9) intangibles" within the meaning of Treasury Regulations Section 1.197-2(h)(1)(i).

- (i) The Company uses the accrual method of accounting for U.S. federal, and all applicable state and local, income Tax purposes. The Company has not ever adopted as a method of accounting, or otherwise accounted for any advance payment or prepaid amount under (i) the "deferral method" of accounting described in Rev. Proc. 2004-34, 2004-22 IRB 991 (or any similar method under state, local or non-U.S. Law) or (ii) the method described in Treasury Regulation Section 1.451-5(b)(1)(ii) (or similar method under state, local or non-U.S. Law).
- (j) The Company is and has been properly treated as an entity that is disregarded as an entity separate from its single owner (Seller) (within the meaning of Treasury Regulation Section 301.7701-2) for federal and applicable state and local income Tax purposes.
- (k) The Company has properly collected and remitted material sales, use, value-added and similar Taxes with respect to sales made to its customers or has properly received and retained any appropriate tax exemption certificates and other documentation for all sales made without charging or remitting sales, use, value-added or similar Taxes that qualify such sales as exempt from sales and similar Taxes.
- (l) The unpaid Taxes of the Company (i) as of the Latest Balance Sheet Date, did not exceed the reserves for Tax liability (excluding any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the balance sheet included in the Financial Statements (and not in related notes and schedules), and (ii) will not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Company in filing its Tax Returns.
- (m) Except as set forth on <u>Schedule 3.18(m)</u>, the Company has (i) not deferred the payment of any "applicable employment taxes" under Section 2302 of the CARES Act, (ii) has not filed for or received any Tax credits under Sections 7001 through 7005 of the Families First Coronavirus Response Act and Section 2301 of the CARES Act, (iii) not deferred any payroll tax obligations pursuant to any Payroll Tax Executive Order, and (iv) not sought (nor has any Affiliate that would be aggregated with the Company and treated as one employer for purposes of Section 2301 of the CARES Act sought) a covered loan under paragraph (36) of Section 7(a) of the Small Business Act (15 U.S.C. 636(a)), as added by Section 1102 of the CARES Act.
- 3.19 <u>Brokerage [***]</u>. Except as set forth on <u>Schedule 3.19</u>, there are no, and will be no, claims for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by this Agreement or the other Transaction Documents based on any arrangement or agreement binding upon Seller Member, Seller or the Company. Except as set forth on <u>Schedule 3.19</u>, there are no [***], special bonuses, severance or other similar compensation (discretionary or otherwise) payable to any current or former officer, director, manager, consultant, contractor or employee of the Company in connection with or arising out of the transactions contemplated hereby.

3.20 Affiliate Transactions.

(a) Schedule 3.20(a) sets forth a true, correct and complete list of each agreement, contract, commitment or transaction between or among (i) the Company, on the one hand, and (ii) (A) Seller or its Affiliates (other than the Company), or (B) any officer, director, manager or employee of the Company or Seller (or any immediate family member of any of the foregoing), on the other hand. Except as set forth on Schedule 3.20(a), no officer, director, manager or employee of the Company or Seller (or any immediate family member of any of the foregoing) owns any beneficial interest in, is a party to any agreement, contract, commitment, or transaction with any members, employees, customers, or suppliers of the

Company or has any right, title or interest in or to any assets or property owned or used or held for use by the Company (including any Intellectual Property Rights).

- (b) Schedule 3.20(b) sets forth a description of all services provided to or on behalf of the Company by Seller and any Affiliates of Seller (including Seller Member) and the costs associated therewith. All of the arrangements set forth on Schedule 3.20(b) are on arms' length terms, including with respect to price. Except as set forth on Schedule 3.20(b), none of the assets, tangible or intangible, or properties that are used by the Company, or that are necessary for the operation of the business of the Company as currently conducted, are owned by Seller or any Affiliate of Seller (other than the Company).
- (c) Except as set forth on <u>Schedule 3.20(c)</u>, since January 1, 2021, all services related to the business of the Company have been provided through the Company, and Seller is not a party to any contracts, agreements or arrangements (written or oral) with any prospective or current customer or vendor of the Company or otherwise related to the business of the Company through another entity or in Seller Member's own capacity.
- (d) Neither Seller nor Seller Member owns or holds, directly or indirectly, any Equity Interests, or the right or obligation to acquire any Equity Interests, or has made any Investment in, or has any right or obligation to make any Investment in, any Person or business (other than the Company), the purpose of which is to conduct business substantially similar to that conducted by the Company or Buyer, other than securities of a publicly traded company which such securities do not exceed more than one percent (1%) of the total voting securities of such company.
- 3.21 <u>Bank Accounts.</u> <u>Schedule 3.21</u> sets forth a true and complete list of (a) the name and address of each bank or other institution at which the Company maintains an account, safe deposit box, lock box or custodial, trust or similar arrangement, (b) the name of each Person authorized to draw funds from, or provide instructions or have access thereto and (c) the account number for each such account of the Company. The Company does not (i) hold or control any funds for any customer or other third party or (ii) provide or arranges for the provision of any payment processing services or otherwise facilitates the transfer of funds for any customer or other third party.

3.22 <u>Anti-Bribery Compliance; International Trade</u>.

- (a) Neither the Company nor any director, manager, officer, direct or indirect equityholder, employee, or agent acting at the direction of the Company has provided, offered, gifted or promised, directly or indirectly, anything of value to any Governmental Authority, political party or candidate for government office, nor provided or promised anything of value to any other person while knowing that all or a portion of that thing of value would or will be offered, given or promised, directly or indirectly, to any Governmental Authority, political party or candidate for government office, for the purpose of:
- (i) influencing any act or decision of such official, party or candidate in his or her official capacity, inducing such official, party or candidate to do or omit to do any act in violation of their lawful duty, or securing any improper advantage for the benefit of the Company; or
- (ii) inducing such official, party or candidate to use his or her influence with his or her government or instrumentality to affect or influence any act or decision of such government or instrumentality, in order to assist the Company in obtaining or retaining business for or with, or directing business to, any person.

- (b) The Company utilizes effective controls procedures and an internal accounting controls system that are sufficient to provide reasonable assurances that violations of applicable anti-bribery or anti-money laundering Laws or regulations will be prevented, detected and deterred.
- (c) The Company has acted at all times within the past five (5) years, in compliance with and without notice of violation of any and all applicable import Laws of any applicable jurisdiction, including without limitation the customs regulations administered by U.S. Customs and Border Protection of the U.S. Department of Homeland Security and the Foreign Trade Regulations administered by the Census Bureau of the U.S. Department of Commerce, as amended from time to time; and in compliance with and without written notice of violation of any required import permits, licenses, authorizations and general licenses granted under such Laws.
- (d) There are no pending or, to the Knowledge of Seller, threatened, legal, regulatory, or judicial requirements or other orders or Action by any Governmental Authority that could have a material effect on the ability or cost to the Company to export, import or sell any items to, from, or in any country (including any pending or threatened tariffs, duties, or other taxes in any country).
- (e) The Company is, and has been since April 24, 2019, in compliance with all U.S. sanctions and export Laws, including the Export Administration Regulations, the International Traffic in Arms Regulations, and the executive orders and regulations administered by the Office of Foreign Assets Control of the U.S. Treasury Department.

3.23 Customer and Suppliers.

- (a) Schedule 3.23(a) sets forth (i) a list of the top five (5) customers of the Company (by volume of sales to such customers) (collectively, the "Material Customers") and (ii) a list of the top five (5) suppliers of the Company (by volume of purchases from such suppliers) (collectively, the "Material Vendor"), in each case, for the fiscal year ended December 31, 2024 and the 10-month period ended October 31, 2025. The Company has not received any written or, to the Knowledge of Seller, oral notice from any Material Customer to the effect that any such Material Customer will stop, decrease the rate of or change the terms (whether related to payment, price or otherwise) with respect to, buying goods, products or services from the Company (whether as a result of the consummation of the transactions contemplated hereby or otherwise). The Company has not received any written or, to the Knowledge of Seller, oral notice from any Material Vendor to the effect that any such Material Vendor will stop, decrease the rate of or change the terms (whether related to payment, price or otherwise) with respect to, supplying materials, products or services to the Company (whether as a result of the consummation of the transactions contemplated hereby or otherwise).
- (b) Except as set forth on <u>Schedule 3.23(b)</u>, there are no claims for indemnification, contribution, reimbursement, or subrogation by or against the Company with respect to any Material Customer or Material Vendor.
- 3.24 <u>Backlog.</u> Attached to <u>Schedule 3.24</u> is a complete and accurate list of the Company's backlog for each project as of October 31, 2025 (the "Backlog"), including the customer name, the project description, aggregate contract value, the revenue recognized as of October 31, 2025 and the remaining contract value. Each of the projects included in the Backlog is currently active and is not, and has never been placed on pause by the applicable customer or the Company.
 - 3.25 <u>Predecessor Entities</u>. The Company does not have any predecessor entities.

3.26 <u>Accounts Receivable</u>. Except as set forth on <u>Schedule 3.26</u>, all accounts receivable of the Company are current and collectible and will be collected in the ordinary course within ninety (90) days.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES CONCERNING BUYER

As a material inducement to Seller and Seller Member to enter into this Agreement and consummate the transactions contemplated hereby, Buyer hereby represents and warrants to Seller and Seller Member as of the Closing that:

- 4.1 <u>Organization and Power</u>. Buyer is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Florida. Buyer has all requisite limited liability company power and authority to execute and deliver this Agreement and the other Transaction Documents to which it is a party, and to perform its respective obligations hereunder and thereunder.
- 4.2 <u>Authorization</u>. The execution, delivery and performance by Buyer of this Agreement and the other Transaction Documents to which Buyer is a party and the consummation of the transactions contemplated hereby and thereby have been duly and validly authorized by Buyer and no other limited liability company act or proceeding on the part of Buyer or its managing member or equityholders is necessary to authorize the execution, delivery or performance of this Agreement and the other Transaction Documents to which Buyer is a party and the consummation of the transactions contemplated hereby and thereby. This Agreement has been duly executed and delivered by Buyer and constitutes a valid and binding obligation of Buyer, enforceable in accordance with its terms, except as may be affected by the Enforceability Limitations, and each of the other Transaction Documents to which Buyer is a party, when executed and delivered by Buyer in accordance with the terms hereof, shall each constitute a valid and binding obligation of Buyer enforceable with its respective terms, except as may be affected by the Enforceability Limitations.
- 4.3 <u>No Violation</u>. Buyer is not subject to nor obligated under its Governing Documents, or any applicable Law, or any material agreement, instrument, license or permit, or subject to any order, writ, injunction or decree, which would be breached or violated by its execution, delivery or performance of this Agreement and the other Transaction Documents to which Buyer is a party or the consummation of the transactions contemplated hereby and thereby.
- 4.4 <u>Governmental Authorities and Consents.</u> No Permit, consent, approval or authorization of, or declaration to or filing with, any Governmental Authority or any other Person is required in connection with the execution, delivery or performance by Buyer of this Agreement and the other Transaction Documents to which Buyer is a party or the consummation by Buyer of the transactions contemplated hereby and thereby.
- 4.5 <u>Litigation</u>. There are no actions, suits, proceedings, orders or investigations pending or, to Buyer's knowledge, threatened against or affecting Buyer, at Law or in equity, or by any Governmental Authority, which would adversely affect Buyer's ability to consummate the Closing.
- 4.6 <u>Brokerage</u>. There are no, and will be no claims for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by this Agreement or the other Transaction Documents based on any arrangement or agreement made by or on behalf of Buyer, other than those which shall be paid by or on behalf of Buyer.

4.7 <u>Investment Purpose</u>. Buyer is acquiring the Purchased Equity solely for its own account for investment purposes and not with a view to, or for offer or sale in connection with, any distribution thereof. Buyer acknowledges that the Purchased Equity is not registered under the Securities Act, as amended, or any state securities laws, and that the Purchased Equity may not be transferred or sold except pursuant to the registration provisions of the Securities Act, as amended, or pursuant to an applicable exemption therefrom and subject to state securities laws and regulations, as applicable. Buyer is able to bear the economic risk of holding the Purchased Equity for an indefinite period (including total loss of its investment), and has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risk of its investment.

ARTICLE V

SURVIVAL: INDEMNIFICATION: OTHER COVENANTS

- 5.1 <u>Survival of Representations and Warranties</u>. The representations and warranties in this Agreement shall survive the Closing as follows:
- (a) the Fundamental Representations shall not terminate hereunder, it being the express intent of the parties hereto to extend the applicable statute of limitations under applicable Law with respect to claims relating to a breach or inaccuracy of any such Fundamental Representations (and the remedies hereunder with respect thereto) for the maximum period permitted under Law (including 10 Del. C. 8106(c));
- (b) the representations and warranties in <u>Section 3.15(a)</u> and <u>Section 3.18</u> shall survive until the applicable statute of limitations plus ninety (90) days; and
- (c) all other representations and warranties shall terminate on the fifteen (15) month anniversary of the Closing Date;

provided that any representation or warranty in respect of which indemnity may be sought under <u>Section 5.2</u>, and the indemnity with respect thereto, shall survive the time at which it would otherwise terminate pursuant to this <u>Section 5.1</u> if notice of the inaccuracy or breach thereof giving rise to such right or potential right of indemnity shall have been given to the party against whom such indemnity may be sought prior to such time (regardless of when the Loss in respect thereof may actually be incurred).

5.2 Indemnification.

- (a) Indemnification by Seller and Seller Member. Subject to the other terms of this Section 5.2, Seller and Seller Member shall, jointly and severally, indemnify the Buyer Indemnitees and save and hold each of them harmless against and pay on behalf of or reimburse such Buyer Indemnitees as and when incurred for any Losses which any such Buyer Indemnitee may suffer, sustain or become subject to, as a result of, in connection with, relating or incidental to or by virtue of:
- (i) any breach of or inaccuracy in any representation or warranty made by Seller in this Agreement;
- (ii) any nonfulfillment, violation or breach of any covenant, agreement or obligation to be performed by Seller under this Agreement;
 - (iii) any Indemnified Taxes;

- (iv) any Closing Indebtedness or Transaction Expenses to the extent not included in the calculation of the Final Purchase Price as determined pursuant to Section 2.3;
- (v) any claim for indemnification under the Governing Documents of the Company arising out of or relating to events or circumstances that occurred prior to the Closing; or
 - (vi) any of the matters set forth on Schedule 5.2(a)(vi) attached hereto.

provided that Seller and Seller Member shall not have any liability under Section 5.2(a)(i) (other than with respect to Fundamental Representations or Losses that arise from or relate to fraud, for which, in either case, no such limitation shall apply) unless and until the aggregate amount of all Losses relating thereto for which Seller would, but for this proviso, be liable exceeds on a cumulative basis an amount equal to \$250,000, after which Seller and Seller Member shall be liable for all such Losses; provided, further that (A) Seller's and Seller Member's aggregate liability under Section 5.2(a)(i) (other than with respect to Fundamental Representations or Losses that arise from or relate to fraud, for which no such limitation shall apply) shall in no event exceed \$3,000,000 and (B) Seller's and Seller Member's aggregate liability under Section 5.2(a)(i) in respect of Fundamental Representations shall not exceed the Base Purchase Price, other than with respect Losses that arise from or relate to fraud, for which no such limitation shall apply.

- (b) Indemnification by Buyer. Subject to the other terms of this Section 5.2, Buyer shall indemnify the Seller Indemnitees and save and hold each of them harmless against and pay on behalf of or reimburse such Seller Indemnitees as and when incurred for any Losses which any such Seller Indemnitee may suffer, sustain or become subject to, as a result of, in connection with, relating or incidental to or by virtue of:
- (i) any breach of or inaccuracy in any representation or warranty made by Buyer in this Agreement; or
- (ii) any nonfulfillment, violation or breach of any covenant, agreement or obligation to be performed by Buyer under this Agreement;

provided that Buyer's aggregate liability under <u>Section 5.2(b)</u> shall not exceed the Base Purchase Price, in each case, other than with respect Losses that arise from or relate to fraud, for which no such limitation shall apply.

Escrow. As long as there are Indemnification Escrow Amounts validly held with (c) the Escrow Agent, any and all Losses payable by Seller or the Seller Member as an Indemnitor pursuant to Section 5.2(a) shall be paid first from the Indemnification Escrow Amount. Within three (3) days following the determination that such payment is due to a Buyer Indemnitee and the delivery of written notice from Buyer of its election to recover such amount from the Indemnification Escrow Amount, Buyer and Seller shall execute and deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release such amount to Buyer. If the Indemnification Escrow Amount is insufficient to cover any and all Losses payable by Seller or the Seller Member as an Indemnitor pursuant to Section 5.2(a), or, in the case of any indemnification of the Seller Indemnitees, the applicable Indemnitor, shall promptly pay such Losses by wire transfer of immediately available funds to an account designated by the applicable Buyer Indemnitee or Seller Indemnitee, as the case may be, within three (3) days after the determination thereof; provided, that the Buyer Indemnitees may elect, in their sole discretion, to setoff all or any portion of such amount owing from Seller or Seller Member against any other amounts otherwise due and payable by any of the Buyer Indemnitees or any of their Affiliates to Seller or Seller Member. Within three (3) Business Days following the twelve (12) month anniversary of the Closing Date (the "Indemnification Escrow Release Date") and Seller's written notice to Buyer that the Indemnification Escrow Release Date has

expired and a balance remains in the escrow account to be returned to Seller, Buyer and Seller shall execute and deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release to Seller any remaining portion of the Indemnification Escrow Amount from the Indemnification Escrow Amount, less any amounts that are subject to pending claims made by any Buyer Indemnitee under this <u>ARTICLE V</u>, until such claims have been satisfied or otherwise resolved at which point Buyer and Seller shall execute and deliver a joint written instruction to the Escrow Agent directing the Escrow Agent to release to Seller any such amount not used to satisfy the indemnification rights of the Buyer Indemnitee under <u>Section 5.2(a)</u>. All indemnification payments under this <u>Section 5.2</u> shall be deemed adjustments to the Purchase Price for U.S. federal and state income Tax purposes.

- (d) Third Party Claims. Any Buyer Indemnitee or Seller Indemnitee making a claim for indemnification under this Section 5.2 (an "Indemnitee") shall notify the indemnifying party (an "Indemnitor") of the claim in writing promptly after receiving written notice of any action, lawsuit, proceeding, investigation or other claim against it (if by a third party), describing the claim, the amount thereof (if known and quantifiable) and the basis thereof; provided that the failure to so notify an Indemnitor shall not relieve the Indemnitor of its obligations hereunder except to the extent that (and only to the extent that) the Indemnitor has been materially prejudiced thereby. Any Indemnitor shall be entitled to participate in the defense of such action, lawsuit, proceeding, investigation or other claim giving rise to an Indemnitee's claim for indemnification at such Indemnitor's expense, and at its option (subject to the limitations set forth below) shall be entitled to assume the defense thereof by appointing a reputable counsel reasonably acceptable to the Indemnitee to be the lead counsel in connection with such defense; provided that, prior to the Indemnitor assuming control of such defense it shall first verify to the Indemnitee in writing that such Indemnitor shall (subject to any dollar limitations set forth herein) be responsible for all liabilities and obligations relating to such claim for indemnification and that (subject to any dollar limitations set forth herein) it shall provide indemnification hereunder to the Indemnitee with respect to such action, lawsuit, proceeding, investigation or other claim; provided, further, that:
- (i) the Indemnitee shall be entitled to participate in the defense of such claim and to employ one counsel of its choice for such purpose; provided that the fees and expenses of such separate counsel shall be borne by the Indemnitee (other than any fees and expenses of such separate counsel that are incurred prior to the date the Indemnitor effectively assumes control of such defense which, notwithstanding the foregoing, shall be borne by the Indemnitor, and except that the Indemnitor shall pay all of the fees and expenses of such separate counsel if the Indemnitee has been advised by counsel that a reasonable likelihood exists of a conflict of interest between the Indemnitor and the Indemnitee);
- (ii) unless otherwise agreed to in writing by the Indemnitee, the Indemnitor shall not be entitled to assume control of such defense and shall pay the fees and expenses of counsel retained by the Indemnitee if (A) the claim for indemnification relates to or arises in connection with any criminal proceeding, action, indictment, allegation or investigation; (B) the claim seeks an injunction or other equitable relief against the Indemnitee; (C) the Indemnitee has been advised by counsel that a reasonable likelihood exists of a conflict of interest between the Indemnitor and the Indemnitee; (D) the Indemnitor failed or is failing to vigorously prosecute or defend such claim; (E) the Indemnitee reasonably believes that the Loss relating to the claim could exceed the maximum amount that such Indemnitee could then be entitled to recover under the applicable provisions of this Section 5.2; or (F) the Indemnitee reasonably believes an adverse determination with respect to the action, lawsuit, investigation, proceeding or other claim giving rise to such claim for indemnification could be detrimental to or injure the Indemnitee's reputation or future business prospects; and
- (iii) if the Indemnitor shall control the defense of any such claim, the Indemnitor shall obtain the prior written consent of the Indemnitee before entering into any settlement of a claim or ceasing to defend such claim if, pursuant to or as a result of such settlement or cessation, injunctive

or other equitable relief or any other restrictions will be imposed against the Indemnitee or if such settlement does not expressly and unconditionally release the Indemnitee from all liabilities and obligations with respect to such claim, without prejudice.

- (e) Direct Claims. Notwithstanding anything herein to the contrary, any claim by an Indemnitee for indemnification not involving a third party claim may be asserted by giving the Indemnitor written notice thereof.
- Limitations. Each of Seller and Seller Member hereby agrees that it shall not (and (f) shall cause its Affiliates not to) make any claim for indemnification against Buyer, the Company, or any of their respective Affiliates by reason of the fact that Seller, Seller Member or any of their respective Affiliates is or was an equityholder, member, director, manager, officer, employee or agent of the Company or its respective Affiliates or is or was serving at the request of the Company or its respective Affiliates as a partner, manager, trustee, director, officer, employee or agent of another entity (whether such claim is for judgments, damages, penalties, fines, costs, amounts paid in settlement, losses, expenses or otherwise and whether such claim is pursuant to any statute, charter document, bylaw, agreement or otherwise) with respect to any proceeding brought by any of the Buyer Indemnitees against Seller or Seller Member pursuant to this Agreement or applicable Law or otherwise, and each of Seller and Seller Member (on its own behalf and on behalf of its Affiliates) hereby acknowledges and agrees that it shall not have any claim or right to contribution or indemnity from the Company or its respective Affiliates with respect to any amounts paid by Seller or Seller Member pursuant to this Agreement. Effective upon the Closing, each of Seller and Seller Member (on its own behalf and on behalf of its Affiliates) hereby irrevocably waives, releases and discharges the Company and its Affiliates from any and all liabilities and obligations to it of any kind or nature whatsoever, whether in its capacity as an equityholder, manager, member, officer, employee, director or agent of the Company or its Affiliates or otherwise (including in respect of any rights of contribution or indemnification), in each case whether absolute or contingent, liquidated or unliquidated, known or unknown, and whether arising under any agreement or understanding (other than this Agreement, the other Transaction Documents and any of the other agreements executed and delivered by Buyer in connection herewith and therewith) or otherwise at Law or in equity, and each of Seller and Seller Member agrees that it shall not (and that it shall cause its Affiliates not to) seek to recover any amounts in connection therewith or thereunder from the Company or its Affiliates. In no event shall the Company or its Affiliates have any liability whatsoever to Seller or Seller Member (or any of their respective Affiliates) for breaches of the representations, warranties, agreements or covenants of Seller, Seller Member or the Company hereunder, and neither Seller nor Seller Member shall (and shall cause its Affiliates not to) in any event seek contribution from the Company or its Affiliates in respect of any payments required to be made by Seller or Seller Member pursuant to this Agreement.
- (g) Materiality. Notwithstanding anything contained herein to the contrary, for purposes of determining whether there has been a breach or inaccuracy in any representation or warranty and the amount of Losses that are the subject matter of a claim for indemnification hereunder, each representation and warranty in this Agreement shall be determined without regard and without giving effect to any Materiality Qualifier set forth in or otherwise applicable to such representation or warranty; provided that no Materiality Qualifier shall be disregarded for such purposes with respect to the representations and warranties set forth in Section 3.7.
- (h) Exclusive Remedy. The remedies provided by this Section 5.2 shall be the sole and exclusive monetary remedies of the Buyer Indemnitees or Seller Indemnitees for the recovery of Losses resulting from, relating to or arising out of this Agreement. Notwithstanding the foregoing or anything else in this Agreement to the contrary, nothing herein shall limit or otherwise affect any Buyer Indemnitee's or Seller Indemnitee's right or ability to make, pursue, enforce or prosecute a claim for (i) equitable relief pursuant to Section 5.6, (ii) fraud, or (iii) breach or violation of the other Transaction Documents.

- 5.3 Press Release and Announcements. Unless required by applicable Law (in which case Buyer, Seller and Seller Member agree to use reasonable efforts to consult with the other parties prior to any such disclosure as to the form and content of such disclosure), no press releases or other public releases of information related to this Agreement or the transactions contemplated hereby will be issued or released without the consent of Buyer and Seller; provided that, without the consent of any other party hereto, (a) Buyer, the Company and their respective Affiliates may issue any such releases of information which do not disclose the economic terms hereof, (b) Buyer and its Affiliates may provide ordinary course communications regarding this Agreement and the transactions contemplated hereby to existing or prospective direct or indirect equityholders on a confidential basis and (c) Buyer, its existing or potential lenders and any of their respective representatives may make public statements or announcements regarding this Agreement and the transactions contemplated hereby as may be necessary in connection with any debt financing being arranged for or on behalf of Buyer.
- 5.4 <u>Use of Name and Information</u>. Each of Seller Member and Seller acknowledges and agrees that, subsequent to the Closing, Buyer and the Company shall exclusively own or have the right to use and license the Company Intellectual Property. Following the Closing, neither Seller nor Seller Member shall, and shall cause their Affiliates not to, use, register or apply for, directly or indirectly, the name "Aquawolf LLC" and any other fictitious or trade names used by the Company, any variations or derivatives thereof, or any names confusingly similar thereto, or any other Company Intellectual Property. Except in furtherance of the provision of services to the Company, each of Seller and Seller Member agrees that it and its Affiliates shall not, directly or indirectly, access, attempt to access, exploit or otherwise use any databases, lists or other records containing information with respect to customers, vendors or other business partners or relationships of the Company.
- 5.5 Expenses. Buyer shall pay all of its fees, costs and expenses incurred in connection with the negotiation of this Agreement and the other Transaction Documents, the performance of its obligations hereunder and thereunder and the consummation of the transactions contemplated hereby and thereby. Seller shall pay all of its and all of its Affiliates' and the Company's fees, costs and expenses incurred in connection with the negotiation of this Agreement and the other Transaction Documents, the performance of its obligations hereunder and thereunder and the consummation of the transactions contemplated hereby and thereby. In addition, other than as provided in Section 5.9(i), Seller shall pay all Closing Indebtedness and Transaction Expenses not otherwise included as a deduction to the Purchase Price on the Closing Statement as finally determined pursuant to Section 2.3.
- 5.6 Specific Performance. Each of the Company, Seller, Seller Member and Buyer acknowledges and agrees that the non-breaching parties would be damaged irreparably in the event any of the provisions of this Agreement is not performed in accordance with its specific terms or is otherwise breached. Accordingly, the parties hereto agree that the other parties shall be entitled to an injunction or injunctions, without the posting of any bond or other security, to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court in the United States or in any state having jurisdiction over the parties and the matter, in addition to any other remedy to which they may be entitled pursuant hereto or otherwise at Law or in equity, and the breaching party waives the defense that an adequate remedy at Law may exist.
- 5.7 <u>Further Assurances</u>. In the event that any further action is necessary or desirable to carry out the purposes of this Agreement and the other Transaction Documents, each of the parties hereto will take such further action (including the execution and delivery of such further instruments and documents) as any other party hereto reasonably may request. Each of Seller Member and Seller shall cooperate, and shall cause its Affiliates to cooperate, with the Company and Buyer following the Closing in any request by the Company or Buyer to transfer, or to cause to be transferred, any right, title or interest in any asset,

tangible or intangible, or property that, as of the Closing Date, is or was used by Company or that is owned by Seller or any of its Affiliates (including Seller Member) or associates.

- 5.8 <u>Confidentiality</u>. Each of Seller and Seller Member agrees not to disclose or use at any time (and shall cause each of its Affiliates not to use or disclose at any time) any Confidential Information. Each of Seller and Seller Member further agrees to take all commercially reasonable steps (and to cause each of its Affiliates to take all commercially reasonable steps) but, in each case, to exercise no less than a reasonable standard of care, to safeguard such Confidential Information and to protect it against disclosure, misuse, espionage, loss and theft. In the event Seller, Seller Member or any of their respective Affiliates is required by Law to disclose any Confidential Information, Seller or Seller Member, as applicable, shall promptly notify Buyer in writing, which notification shall include the nature of the legal requirement and the extent of the required disclosure, and Seller or Seller Member, as applicable, shall cooperate with Buyer and the Company to preserve the confidentiality of such information consistent with applicable Law.
- 5.9 <u>Tax Matters</u>. The following provisions shall govern the allocation of responsibility as between Buyer and Seller for certain tax matters following the Closing Date:
- (a) In the case of any taxable period that includes (but does not end on) the Closing Date (a "Straddle Period"), (i) the amount of any Taxes other than real and personal property Taxes (and similar Taxes imposed solely on a periodic basis) of the Company for the portion of the Straddle Period ending on the Closing Date shall be determined based on an interim closing of the books as of the close of business on the Closing Date, except that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions), other than with respect to property placed in service after the Closing, shall be allocated on a daily basis, and (ii) the amount of real and personal property Taxes (and similar Taxes imposed solely on a periodic basis) of the Company for the portion of the Straddle Period ending on the Closing Date shall be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction the numerator of which is the number of days in the portion of the Straddle Period ending on the Closing Date and the denominator of which is the number of days in such Straddle Period.
- (b) Buyer shall cause the Company to prepare and timely file all Tax Returns of the Company due after the Closing Date or overdue as of the Closing Date (the "Buyer Prepared Returns"). With respect to any Buyer Prepared Return that either shows an Indemnified Tax or relates to a Straddle Period, not less than fifteen (15) days prior to the due date for filing any Tax Return for a Straddle Period (taking into account any valid, applicable extensions), but in no event earlier than twenty (20) days after the Closing Date, Buyer shall deliver a copy of such Tax Returns to Seller for its review and comment thereon. Buyer shall cause the Company to incorporate any reasonable comments made by Seller within five (5) days of receipt of such draft Buyer Prepared Return in the Buyer Prepared Return actually filed.
- (c) Seller shall, at Seller's cost and expense, prepare or cause to be prepared and file or cause to be filed all Pass-Through Tax Returns required to be filed by the Company with respect to any taxable period ending on or prior to the Closing Date and the initial due date of which is after the Closing Date. All such Pass-Through Tax Returns shall be prepared and filed in a manner that is consistent with the prior practices, procedures, and accounting methods of the Company, except as otherwise contemplated by this Agreement or required in order to be supportable at a "more likely than not" (or higher) confidence level under applicable Law.
- (d) Buyer, the Company and Seller shall cooperate fully, as and to the extent reasonably requested by the other party, in connection with the filing of Tax Returns pursuant to this <u>Section 5.9</u> and any audit, litigation or other proceeding with respect to Taxes. Such cooperation shall include the retention and (upon the other party's request) the provision of records and information which are reasonably

relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. The Company and Seller agree to retain all books and records in their possession with respect to Tax matters pertinent to the Company relating to any Pre-Closing Tax Period until the expiration of the applicable statute of limitations (and, to the extent notified by Buyer or Seller, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority.

- (e) Buyer or Seller, as the case may be, shall provide notice to the other party upon receipt of any written notice from a Tax authority of a Tax audit, dispute or other Tax proceeding with respect to the Company for a Pre-Closing Tax Period or Straddle Period, if such Tax audit, dispute or other Tax proceeding relates to a Pass-Through Tax Return (a "Covered Tax Proceeding").
- (i) Within ten (10) days of the receipt of notice from the Buyer of a Covered Tax Proceeding, Seller may elect to control, at Seller's expense, the contest or resolution of such Covered Tax Proceeding unless such Covered Tax Proceeding relates to a Straddle Period; provided that Seller shall (A) keep Buyer reasonably informed with respect to the status of such Covered Tax Proceeding, including by giving Buyer advance notice of, and opportunity to attend, any material in-person or telephonic meetings, (B) provide copies to Buyer of any material written correspondence or other material submissions received from a Tax authority with respect to such Covered Tax Proceeding, (C) provide copies of any material written correspondence to be provided to any Tax authority in connection with such Covered Tax Proceeding to Buyer for Buyer's review and comment, with all reasonable comments of Buyer to be reflected in such submission, and (D) provide to Buyer (at Buyer's sole cost and expense) reasonable participation rights with respect to any such Covered Tax Proceeding; provided, further, that Seller shall not enter into any settlement of, or otherwise compromise, any such Covered Tax Proceeding without the prior written consent of the Buyer (which consent shall not be unreasonably withheld, conditioned or delayed).
- (ii) In the case of a Covered Tax Proceeding (A) which Seller does not timely elect to control pursuant to the preceding sentence and/or (B) which relates to a Straddle Period, Buyer shall control such Covered Tax Proceeding, provided that the Buyer shall (1) keep Seller reasonably informed with respect to the status of such Covered Tax Proceeding, including by giving Seller advance notice of, and opportunity to attend, any material in-person or telephonic meetings, (2) provide copies to Buyer of any material written correspondence or other material submissions received from a taxing authority with respect to such Covered Tax Proceeding, (3) provide copies of any material written correspondence to be provided to any Tax authority in connection with such Covered Tax Proceeding to Seller for Seller's review and comment, with all reasonable comments of Seller to be reflected in such submission, and (4) provide to Seller (at Seller's sole cost and expense) reasonable participation rights with respect to any such Covered Tax Proceeding; provided, further that Buyer shall not enter into any settlement of, or otherwise compromise, any such Covered Tax Proceeding without the prior written consent of Seller (which consent shall not be unreasonably withheld, conditioned or delayed).
- (iii) To the extent that there is any inconsistency between <u>Section 5.2(d)</u> or <u>Section 5.2(e)</u> and this <u>Section 5.9(e)</u> as it relates to any matters relating to Taxes, the provisions of this <u>Section 5.9(e)</u> shall govern.
- (f) All Tax sharing agreements or similar agreements with respect to or involving the Company shall be terminated as of the Closing Date and, after the Closing Date, the Company shall not be bound thereby or have any liability thereunder.
- (g) The parties hereby agree and acknowledge that the proposed acquisition of the Purchased Equity is intended to be treated as a purchase and sale of an undivided interest in the assets of

the Company for federal and applicable state Income Tax purposes, and each of the parties agrees to file their respective Tax Returns in a manner consistent with such intended Tax treatment.

- The consideration paid by Buyer for the Purchased Equity, plus any liabilities (h) assumed or deemed assumed by Buyer and any other amounts treated as consideration for Income Tax purposes, in each case, with respect to the Purchased Equity, shall be allocated among the assets of the Company for Income Tax purposes in accordance with Section 1060 of the Code and the Treasury Regulations thereunder (and any similar provisions of state, local or non-U.S. Law as appropriate) and pursuant to the estimated allocation prepared by Seller and Buyer set forth on Schedule I (as may be adjusted as contemplated in this Section 5.9(h), the "Allocation Statement"). Notwithstanding this Section 5.9(h), the parties hereto agree that the allocation pursuant to the Allocation Statement shall be further adjusted to reflect any indemnification payments made pursuant to ARTICLE V that are treated as adjustments to the purchase price pursuant to Section 5.2(c), the release of any portion of the Indemnification Escrow Amount that is paid to Seller pursuant to Section 5.2(c), and any other adjustments after the Closing to the Purchase Price (and any other amounts treated as consideration for U.S. federal and applicable state and local Tax purposes) in a manner consistent with the allocation agreed upon pursuant to this Section 5.9(h) and Section 1060 of the Code and the Treasury Regulations thereunder. Each of the parties hereto and their respective Affiliates shall, unless otherwise required by a final "determination" (within the meaning of Section 1313(a) of the Code), (A) prepare and file all Tax Returns, including all IRS Forms 8594, in a manner consistent with the Allocation Statement, and (B) take no position in any Tax Return, tax proceeding or otherwise that is inconsistent with the Allocation Statement, in each case, as finally determined pursuant to this Section 5.9(h). In the event that any of the allocations set forth in the Allocation Statement are disputed by any Taxing Authority, the party receiving notice of such dispute shall promptly notify and consult with the other party concerning the resolution of such dispute.
- (i) All transfer, documentary, sales, use, stamp, registration and other such Taxes, and all conveyance fees, recording charges and other fees and charges (including any penalties and interest) ("Transfer Taxes") incurred in connection with the consummation of the transactions contemplated by this Agreement shall be borne one-half by Buyer and one-half by Seller, and the party required by applicable Law to file any Tax Return or other document with respect to such Transfer Taxes shall file, or cause to be filed, all necessary Tax Returns and other documentation with respect to all such Transfer Taxes, and, if required by applicable Law, the other party shall join in the execution of any such Tax Returns and other documentation.
- 5.10 Release. Seller and Seller Member (on each of their own behalf and on behalf of their respective former and current Affiliates and its and their respective former and current officers, directors, employees, agents, representatives, members, managers, partners, and equityholders, and all of their respective predecessors, successors, heirs and assigns (collectively, the "Releasors")) hereby unconditionally and irrevocably acquits, remises, waives, releases and forever discharges the Company (and its predecessors) and their respective former and current Affiliates (including New Mountain Capital, L.L.C.) and their respective officers, directors, employees, managers, agents, insurers, attorneys, members, representatives, predecessors, successors and assigns (collectively, the "Releasees") from any and all liabilities, losses and obligations of every kind or nature whatsoever, whether in any Releasor's capacity as an equityholder, creditor, lender, director, manager, officer, employee or agent of any Releasee or otherwise, whether accrued or fixed, liquidated or unliquidated, known or unknown, suspected or unsuspected, absolute or contingent, matured or unmatured, or determined or determinable, including those arising under any Law, contract, arrangement, commitment, transaction or undertaking, whether written or oral, to the extent arising prior to the Closing (except for claims under this Agreement) (the "Released Claims"). Each of Seller and Seller Member understands that this general release shall apply to all unknown or unanticipated claims, damages and causes of action, as well as those known and anticipated. Notwithstanding the foregoing, the Released Claims shall not include, and the foregoing release of claims

shall not apply to claims that cannot be waived by private agreement under applicable Law. Each of Seller and Seller Member (on each such party's own behalf and on behalf of the other Releasors) agrees that, from and after the Closing, the Releasors shall not seek to recover any amounts in connection with any Released Claim from any Releasee. In furtherance of the foregoing, each of Seller and Seller Member, on such party's own behalf and on behalf of the other Releasors, hereby waives and agrees not to assert any rights under any Law to the effect that a general release does not extend to claims that the creditor does not know or suspect at the time of executing the release, including Section 1542 of the California Civil Code. Each of Seller and Seller Member agrees that in no event shall the Company or its Affiliates have any liability to Seller or Seller Member, and neither Seller nor Seller Member shall have any right to contribution or indemnity from the Company or its Affiliates, with respect to any amounts paid or payable by or indemnification obligations of Seller or Seller Member pursuant to this Agreement or for breaches of the representations, warranties, agreements, or covenants of Seller or Seller Member hereunder.

5.11 Restrictive Covenants.

- Each of Seller and Seller Member (each a "Restricted Party" and collectively "Restricted Parties") acknowledges that an important part of the benefit that Buyer will receive in connection with the transactions contemplated hereby is the ability to conduct the business of the Company free from competition from Restricted Parties. In order that Buyer may enjoy such benefits, each Restricted Party agrees that for the Non-Compete Restricted Period (in the case of the following clauses (i) and (iv)) and the Non-Solicit Restricted Period (in the case of the following clauses (ii) and (iii)), each Restricted Party will not, and each will cause their respective Affiliates not to, directly or indirectly: (i) engage or participate in, have any ownership or other interest (whether financial or otherwise) in, or in any way assist (as an employee, agent, consultant, investor, partner, shareholder, member, lender, director, officer, or otherwise) any Person to engage in the operation of the Business, any other business engaged in by the Company prior to the Closing, or a business substantially similar to the Business or to such other business, or any portion thereof, anywhere in the United States; (ii) attempt to induce or encourage others to induce any employee of or consultant to the Company or any of its Affiliates, to terminate such person's employment with the Company or any of its Affiliates (in the case of an employee) or cease providing services to the Company or any of its Affiliates (in the case of a consultant); (iii) solicit or hire or encourage any Person to solicit or hire any employee of the Company or any of its Affiliates or any person who was an employee of the Company or any of its Affiliates during the one (1) year preceding the date of proposed hire; or (iv) divert, solicit, or interfere with, or attempt to divert, solicit, or interfere with, or assist or encourage any Person in diverting, soliciting, interfering with, or attempting to divert, solicit, or interfere with, any customer, client, or supplier, or prospective customer, client, or supplier, of the Company or any of its Affiliates. Notwithstanding the foregoing, the provisions of this Section 5.11(a) will not prohibit Restricted Parties from being a passive owner of not more than one percent (1%) of the outstanding stock of any class of a corporation that is listed on a national securities exchange.
- (b) Each Restricted Party will not, and each will cause its respective Affiliates not to, directly or indirectly, make any negative statement or communication regarding Buyer, the Company, or any of their respective Affiliates, or their respective officers, directors, managers, shareholders, members, equityholders, partners, employees, contractors, agents, successors, or assigns or any of their products, services, or businesses; provided, however, that neither Restricted Party nor any of its respective Affiliates will be prevented from making true and accurate statements or communications in connection with any disclosure to any government agency or that they reasonably believe are required pursuant to applicable Law or necessary to enforce any rights under this Agreement.
- (c) Each Restricted Party agrees and acknowledges that the duration and scope of the restrictive covenants contained in this <u>Section 5.11</u> are fair, reasonable, and necessary in order to protect the legitimate interests of Buyer, and that adequate consideration has been received for such obligations by

Restricted Parties. If, however, for any reason any court determines that the restrictions in this <u>Section 5.11</u> are not reasonable or that such consideration is inadequate, such restrictions will be interpreted, modified, or rewritten to include as much of the duration, scope, and geographic area identified in this <u>Section 5.11</u> as will render such restrictions valid and enforceable.

- (d) Each Restricted Party acknowledges that any breach or threatened breach of the provisions contained in this Section 5.11 will result in serious and irreparable injury to Buyer. Therefore, each Restricted Party acknowledges and agrees that in the event of a breach or threatened breach by either Restricted Party or any of its respective Affiliates, Buyer will be entitled, in addition to any other remedy at Law or in equity to which Buyer may be entitled, to equitable relief against such Person, including an injunction to restrain such Person from such breach and to compel compliance with the obligations of such Person hereunder in protecting or enforcing Buyer's rights and remedies, all without the posting of any bond or other security.
- 5.12 Insurance Matters. To the extent that Buyer, any of its Affiliates or the Company has a Proceeding or insurance claim that is related to the Company based on any actual or alleged fact, action, failure to act, injury, damage or circumstance existing or occurring prior to the Closing Date under any insurance policies and related umbrella and excess policies of Seller Member, including (a) commercial general liability or general liability insurance policy, (b) umbrella liability insurance policy, (c) property, (d) workers compensation and employers liability policy and (e) automobile liability policy (collectively, the "Seller Member Insurance Policies"), the Company's rights to coverage under such policies will continue with respect to such Proceedings and insurance claims without the assessment of any charges or charge-back fees or premium to Buyer or any of its Affiliates or any requirement that such parties make any payment with respect thereto or reimburse Seller Member or any insurers with respect thereto. After the Closing, Seller and Seller Member shall, in good faith, timely submit on behalf of Buyer, its Affiliates or the Company all claims under the Seller Member Insurance Policies, and claims for Losses relating to, arising out of, or based on any fact, action, failure to act, injury, damage or circumstance existing or occurring prior to the Closing Date relating to the Company for which insurance coverage is reasonably likely to be available under the Seller Member Insurance Policies (each such claim, an "Insurance Claim"). After the Closing, Seller and Seller Member shall timely submit such Insurance Claims on behalf of Buyer, its Affiliates or the Company for adjustment and coverage in accordance with the Seller Member Insurance Policies to the same extent as Seller, Seller Member and other Persons with respect to the Company were entitled prior to the Closing. Seller and Seller Member shall deliver promptly any payments received with respect to any such Insurance Claim to Buyer or as otherwise directed in writing by Buyer. Seller and Seller Member shall provide Buyer and its Affiliates with reasonable cooperation regarding the adjustment and pursuit of coverage for each Insurance Claim and the Buyer and its Affiliates shall have the right to effectively associate in the adjustment and pursuit of coverage for any Insurance Claim. Seller Member shall provide Buyer with thirty (30) days' prior written notice of cancellation of the Seller Member Insurance Policies.

ARTICLE VI

MISCELLANEOUS

Agreement may be waived; provided that any such amendment or waiver will be binding upon Seller and Seller Member only if such amendment or waiver is set forth in a writing executed by Seller and any such amendment or waiver will be binding upon the Company or Buyer only if such amendment or waiver is set forth in a writing executed by the Company or Buyer, respectively. No course of dealing between or among any Persons having any interest in this Agreement shall be deemed effective to modify, amend or waive any part of this Agreement or any rights or obligations of any Person under or by reason of this Agreement.

No waiver of any of the provisions of this Agreement shall be effective unless in writing and no waiver shall be deemed to, nor shall any waiver, constitute a waiver of any other provisions, regardless of whether similar, nor shall any waiver constitute a continuing waiver.

6.2 <u>Notices</u>. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be deemed to have been given (a) if personally delivered, on the date of delivery, (b) if delivered by next day courier service of national standing (with charges prepaid), on the Business Day following the date of delivery to such courier service for next day delivery, (c) if deposited in the United States mail, first class postage prepaid, on the fifth (5th) Business Day following the date of such deposit, or (d) if sent by email, (i) on the date sent, if sent on a Business Day, and (ii) on the next Business Day following the date sent, if sent on a non-Business Day. Notices, demands and communications shall, unless another address is specified in writing pursuant to the provisions hereof, be sent to the address indicated below:

Notices to Seller and/or Seller Member:

Mammoth Energy Partners LLC Mammoth Energy Services, Inc. 14201 Caliber Drive, Suite 300 Oklahoma City, OK 73134

Email: [***]

Attention: Mark Layton

with a copy (which shall not constitute notice) to:

Hall, Estill, Hardwick, Gable, Golden & Nelson, P.C. 521 East 2nd Street, Suite 1200 Tulsa, OK 74120 Email: [***]; [***] Attention: Stephen W. Ray and Betsy G. Jackson

Qualus, LLC 100 Colonial Center Parkway, Suite 400 Lake Mary, FL 32746

Notices to the Company and/or Buyer:

Email: [***]; and [***]
Attention: Patrick Curran and Candice Mountz

with a copy (which shall not constitute notice) to: New Mountain Capital 1633 Broadway, 48th Floor New York, NY 10019 Email: [***] and [***]

Attention: Rishi Abuwala and Zach Levy

with a copy (which shall not constitute notice) to:

Lowenstein Sandler LLP
1251 Avenue of the Americas
New York, NY 10020
Attention: David Goret; Michael Reinhardt
Email: [***]; [***]

- 6.3 Successors and Assigns. This Agreement and all of the covenants and agreements contained herein and rights, interests or obligations hereunder, by or on behalf of any of the parties hereto, shall bind and inure to the benefit of the respective heirs, successors and permitted assigns of the parties hereto, except that no party may assign, delegate or otherwise transfer either this Agreement or any of the covenants and agreements herein or rights, interests or obligations hereunder, without the prior written consent of Buyer and Seller. Notwithstanding the foregoing, Buyer may assign its rights and obligations hereunder, in whole or in part, to (a) any of its Affiliates, (b) in connection with any disposition or transfer of all or any portion of the Company or its businesses in any form of transaction or (c) to any of its lenders as collateral security, in each case, without the consent of any of the other parties hereto; provided, however, that, with respect to any assignments pursuant to the foregoing clause (a) Buyer remains liable for the full and timely performance of all of Buyer's obligations under this Agreement.
- 6.4 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Agreement.

- 6.5 Interpretation. The headings and captions used in this Agreement, in any Schedule or Exhibit hereto, in the table of contents or in any index hereto are for convenience of reference only and do not constitute a part of this Agreement and shall not be deemed to limit, characterize or in any way affect any provision of this Agreement or any Schedule or Exhibit hereto, and all provisions of this Agreement and the Schedules and Exhibits hereto shall be enforced and construed as if no caption or heading had been used herein or therein. Any capitalized terms used in any Schedule or Exhibit attached hereto and not otherwise defined therein shall have the meanings set forth in this Agreement. Each defined term used in this Agreement shall have a comparable meaning when used in its plural or singular form. The words such as "herein," "hereinafter," "hereof" and "hereunder" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires. The use of the word "including" (or deviations thereof) herein shall mean "including without limitation." The phrase "made available" or "disclosed" and phrases of similar import shall mean, with respect to any information, document or other material, that such information, document or material was made available by Seller or any of its Representatives in the "Project Puma" virtual data room hosted by Ideals and established in connection with the transactions contemplated by this Agreement at least two (2) Business Days prior to the date of this Agreement and remained available for review by Buyer and its Representatives through the date hereof. The phrase "ordinary course of business" or phrase of similar import shall mean, with respect to the applicable Person, the usual and ordinary course of such Person's business consistent with past custom and practice (including with respect to frequency, quantity and magnitude). The parties hereto intend that each representation, warranty and covenant contained herein shall have independent significance. If any party hereto has breached any representation, warranty or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which such party has not breached shall not detract from or mitigate the fact that such party is in breach of the first representation, warranty or covenant. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.
- 6.6 No Third Party Beneficiaries. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any Person other than the parties hereto and their respective permitted successors and assigns, any rights or remedies under or by reason of this Agreement, such third parties specifically including employees and creditors of the Company. Notwithstanding the foregoing, (a) the Buyer Indemnitees are express beneficiaries of Section 5.2, and (b) the Non-Recourse Parties are express third-party beneficiaries of Section 6.12.
- 6.7 <u>Complete Agreement</u>. This Agreement and the agreements and documents referred to herein contain the entire agreement and understanding among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings whether written or oral, relating to such subject matter in any way, including that certain letter of intent, dated as of May 9, 2025.
- 6.8 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which taken together shall constitute one and the same instrument.
- 6.9 <u>Electronic Delivery</u>. This Agreement and any signed agreement or instrument entered into in connection with this Agreement, and any amendments hereto or thereto, to the extent delivered by means of email in portable document format or similar format (any such delivery, an "Electronic Delivery"), shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto or to any such agreement or instrument, each other party hereto or thereto shall re-execute original forms thereof and deliver them (by means other than Electronic Delivery) to all other

parties. No party hereto or to any such agreement or instrument shall raise the use of Electronic Delivery to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of Electronic Delivery, as a defense to the formation of a contract, and each such party forever waives any such defense, except to the extent such defense related to lack of authenticity.

6.10 Governing Law; JURISDICTION; WAIVER OF JURY TRIAL.

- (a) This Agreement shall be interpreted and construed in accordance with the Laws of the State of Delaware. Any and all claims, controversies, and causes of action arising out of or relating to this Agreement, whether sounding in contract, tort, or statute, shall be governed by the Laws of the State of Delaware, including its statutes of limitations, without giving effect to any conflict-of-laws or other rule that would result in the application of the Laws of a different jurisdiction.
- (b) SUBJECT TO THE PROVISIONS OF <u>SECTION 2.3</u> (WHICH SHALL GOVERN ANY DISPUTE ARISING HEREUNDER), each party hereto agrees that it shall bring any action between the parties hereto arising out of or related to this Agreement exclusively in the Court of Chancery of the State of Delaware or the Complex Commercial Litigation Division for the Superior Court for New Castle County, Delaware, as applicable, and any state appellate court therefrom within the State of Delaware (or, if either the Chancery Court of the State of Delaware or the Complex Commercial Litigation Division for the Superior Court for New Castle County, as applicable, declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), and each party waives any objection which such party may now or hereafter have to the laying of the venue of any such action in such court, and irrevocably submits to the exclusive jurisdiction of any such court referenced in this Section 6.10(b) in any such action and waives any objection that any such court is an inconvenient forum or does not have jurisdiction over any party hereto. Each party agrees that service of process upon such party in any such action shall be effective if notice is given in accordance with Section 6.2.
- (c) THE PARTIES TO THIS AGREEMENT EACH HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (I) ARISING UNDER THIS AGREEMENT OR (II) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO IN RESPECT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS RELATED HERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. THE PARTIES TO THIS AGREEMENT EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT THE PARTIES TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.
- 6.11 <u>Schedules</u>. No exceptions to any representations or warranties disclosed on one Schedule shall constitute an exception to any other representations or warranties made in this Agreement unless the exception is disclosed as provided herein on each such other applicable Schedule or unless the relevance of such exception to such other representation and warranty is reasonably apparent on its face based upon a plain reading of such exception; provided that in no event shall a disclosed exception on any section or subsection of any Schedule be deemed to qualify the representations and warranties set forth in <u>Section 3.7</u>, unless such exception is specifically set forth on <u>Schedule 3.8</u>.

6.12 Non-recourse.

- (a) This Agreement may only be enforced against, and any claim, action, suit or other legal proceeding based upon, arising out of, or related to this Agreement, or the negotiation, execution or performance of this Agreement, may only be brought against the parties hereto and then only with respect to the specific obligations set forth herein with respect to such party. Except as otherwise provided in this Agreement or the other Transaction Documents, no past, present or future director, officer, employee, incorporator, manager, member, partner, stockholder, Affiliate, agent, attorney or other Representative of any party hereto or of any Affiliate of any party hereto (including New Mountain Capital, L.L.C.), or any of their successors or permitted assigns (collectively, the "Non-Recourse Parties"), shall have any liability for any obligations or liabilities of any party hereto under this Agreement or for any claim, action, suit or other legal proceeding based on, in respect of or by reason of the transactions contemplated hereby. Notwithstanding the foregoing, nothing herein shall limit, restrict or otherwise affect a Person's rights or obligations under any other Transaction Document.
- (b) The provisions of this Section 6.12, together with the remedies specified in Section 5.2, were specifically bargained-for between Seller and Buyer and were taken into account by Seller and Buyer in agreeing to the amount of the Purchase Price, the adjustments thereto and the other terms and conditions hereof. Seller, the Company and Buyer have specifically relied upon the provisions of this Section 6.12 together with the remedies specified in Section 5.2, in agreeing to the Purchase Price, the adjustments thereto and the other terms and conditions hereof, including in agreeing to provide the specific representations and warranties set forth herein.
- 6.13 <u>Time Periods</u>. Unless specified otherwise, any action required hereunder to be taken within a certain number of days shall be taken within that number of calendar days (and not Business Days); provided, however, that if the last day for taking such action falls on a day that is not a Business Day, the period during which such action may be taken shall be automatically extended to the next Business Day.

* * * * *

IN WITNESS WHEREOF, the parties hereto have executed this Equity Purchase Agreement on the date first written above.

COMPANY:

AQUAWOLF LLC

By: <u>/s/ Mark Layton</u> Name: Mark Layton Its: Chief Financial Officer

SELLER:

MAMMOTH ENERGY PARTNERS LLC

By: /s/ Mark Layton Name: Mark Layton

Its: Chief Financial Officer

SELLER MEMBER:

MAMMOTH ENERGY SERVICES, INC.

By: /s/ Mark Layton Name: Mark Layton

Its: Chief Financial Officer

BUYER:

QUALUS, LLC

By: /s/ Candice Mountz Name: Candice Mountz Its: General Counsel

[Signature Page to Equity Purchase Agreement]



Mammoth Energy Services, Inc. Announces Sale of Engineering Business

Sales Price of \$30.0 million; Further Expands Deployment Opportunities

OKLAHOMA CITY - Dec. 2, 2025 - Mammoth Energy Services, Inc. (NASDAQ: TUSK) ("Mammoth" or the "Company") today announced that its subsidiary, Mammoth Energy Partners LLC, has completed a sale of all equity interests in its wholly owned subsidiary, Aquawolf LLC ("Aquawolf"), to Qualus, LLC for an aggregate sales price of \$30.0 million, advancing the Company's ongoing transformation and portfolio optimization initiatives.

Aquawolf's revenue grew from \$12.1 million in 2022 to \$17.3 million in 2024, with net income rising from \$1.1 million to \$1.8 million over the same period. For the nine months ended September 30, 2025, Aquawolf generated \$12.0 million in revenue and \$1.3 million in net income.

At closing, Mammoth Energy Partners LLC received total cash proceeds of \$23.5 million. An additional \$2.5 million was placed into escrow to fund post-closing adjustments for at least ninety days and indemnified liabilities until at least December 1, 2026.

In conjunction with the transaction described above, the Company entered into a consent and release agreement with Fifth Third Bank, National Association ("Fifth Third"), in which Fifth Third consented to the transaction and agreed to release associated collateral. The Company's borrowing base remains unchanged.

Mark Layton, Mammoth's Chief Financial Officer, stated, "This transaction provides another clear proof point of the strategic value we've unlocked this year. We entered the Engineering business in 2018 with one manager and a small initial investment, and over time built it into a team generating \$17.3 million of revenue in 2024. Monetizing this business for approximately \$30.0 million is a strong outcome and a credit to the people who built the business.

More importantly, this sale reinforces our belief that the underlying value across Mammoth is significantly disconnected from the current share price. We've shown an ability to build and monetize businesses organically, and we'll continue to deploy capital where it creates returns and supports long-term value for shareholders."

About Mammoth Energy Services, Inc.

Mammoth is an integrated, growth-oriented company providing a diversified suite of rental, infrastructure and energy services across North America. Mammoth's offerings span specialized equipment rentals supporting aviation, construction, and energy operations, as well as utility infrastructure solutions, including engineering, design and fiber optic services. The Company also provides natural sand proppant for hydraulic fracturing, directional drilling services, and workforce accommodation facilities designed to support large-scale projects in remote locations. By combining technical expertise with a broad service platform, Mammoth helps customers achieve greater efficiency, flexibility and value across their operations. For more information, please visit www.mammothenergy.com.

Forward-Looking Statements and Cautionary Statements

This news release (and any oral statements made regarding the subjects of this release, including on the conference call announced herein) contains certain statements and information that may constitute "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995. All statements, other than statements of historical facts that address activities, events or developments that Mammoth expects, believes or anticipates will or may occur in the future are forward-looking statements. The words "anticipate," "believe," "ensure," "expect," "if," "intend," "plan," "estimate," "project,"

"forecasts," "predict," "outlook," "aim," "will," "could," "should," "potential," "would," "may," "probable," "likely" and similar expressions, and the negative thereof, are intended to identify forward-looking statements. Without limiting the generality of the foregoing, forward-looking statements contained in this news release specifically include statements, estimates and projections regarding the Company's business outlook and plans, future financial position, liquidity and capital resources, operations, performance, acquisitions, returns, capital expenditure budgets, plans for stock repurchases under its stock repurchase program, costs and other guidance regarding future developments. Forward-looking statements are not assurances of future performance. These forward-looking statements are based on management's current expectations and beliefs, forecasts for the Company's existing operations, experience and perception of historical trends, current conditions, anticipated future developments and their effect on Mammoth, and other factors believed to be appropriate. Although management believes that the expectations and assumptions reflected in these forward-looking statements are reasonable as and when made, no assurance can be given that these assumptions are accurate or that any of these expectations will be achieved (in full or at all). Moreover, the Company's forward-looking statements are subject to significant risks and uncertainties, including those described in its Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and other filings it makes with the SEC, including those relating to the Company's acquisitions and contracts, many of which are beyond the Company's control, which may cause actual results to differ materially from historical experience and present expectations or projections which are implied or expressed by the forward-looking statements. Important factors that could cause actual results to differ materially from those in the forward-looking statements include, but are not limited to: demand for our services; the volatility of oil and natural gas prices and actions by OPEC members and other exporting nations affecting commodities prices and production levels; conditions of U.S. oil and natural gas industry and the effect of U.S. energy, monetary and trade policies; U.S. and global economic conditions and political and economic developments, including the energy and environmental policies; changes in U.S. and foreign trade regulations and tariffs, including potential increases of tariffs on goods imported into the U.S., and uncertainty regarding the same; performance of contracts and supply chain disruptions; inflationary pressures; higher interest rates and their impact on the cost of capital; the failure to receive or delays in receiving the remaining payment under the settlement agreement with PREPA; the Company's inability to replace the prior levels of work in its business segments, including its well completion services segments; risks relating to economic conditions, including concerns over a potential economic slowdown or recession; the loss of or interruption in operations of one or more of Mammoth's significant suppliers or customers; the outcome or settlement of our litigation matters and the effect on our financial condition and results of operations; the effects of government regulation, permitting and other legal requirements; operating risks; the adequacy of capital resources and liquidity; Mammoth's ability to comply with the applicable financial covenants and other terms and conditions under its revolving credit facility; weather; natural disasters; litigation; volatility in commodity markets; competition in the oil and natural gas and infrastructure industries; and costs and availability of resources.

Investors are cautioned not to place undue reliance on any forward-looking statement which speaks only as of the date on which such statement is made. We undertake no obligation to correct, revise or update any forward-looking statement after the date such statement is made, whether as a result of new information, future events or otherwise, except as required by applicable law.

MAMMOTH ENERGY SERVICES, INC. UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

On December 2, 2025, Mammoth Energy Partners LLC ("MEP"), a subsidiary of Mammoth Energy Services, Inc. ("Mammoth" or the "Company"), entered into an Equity Purchase Agreement (the "Agreement"), as the seller, with Qualus, LLC ("Qualus"), as the buyer, and Aquawolf LLC ("Aquawolf"), MEP's wholly-owned subsidiary and subject of the sale, as a party to the Agreement. Pursuant to the Agreement, MEP sold all equity interests in Aquawolf (the "Engineering Business"), which was included in the Company's Infrastructure segment, to Qualus for \$30.0 million ("Transaction A"). Transaction A was completed simultaneously with the signing of the Agreement on December 2, 2025. The Company will report the results of its Engineering Business as discontinued operations in the Company's consolidated financial statements beginning in its Annual Report on Form 10-K for the annual period ending December 31, 2025.

As previously reported, on June 16, 2025, Stingray Pressure Pumping LLC ("Stingray") and Mammoth Equipment Leasing LLC ("Mammoth Equipment"), subsidiaries of Mammoth Energy Services, Inc. ("Mammoth" or the "Company"), entered into an Equipment Purchase Agreement, as the sellers, with MGB Manufacturing, LLC ("MGB"), as the buyer, pursuant to which Stingray and Mammoth Equipment sold all of its equipment used in its hydraulic fracturing business, which is included in the Company's Well Completion segment, to MGB for \$15.0 million ("Transaction B"). In addition, on April 11, 2025, Lion Power Services LLC ("Lion"), a subsidiary of Mammoth, entered into an Equity Purchase Agreement, as the seller, with Peak Utility Services Group, Inc., as the buyer, pursuant to which Lion sold all equity interests in its wholly-owned subsidiaries 5 Star Electric, LLC ("5 Star"), Higher Power Electrical, LLC ("Higher Power") and Python Equipment LLC ("Python" and collectively with 5 Star and Higher Power, the "T&D Business") ("Transaction C"). The aggregate sales price in connection with the Transaction was approximately \$108.7 million, subject to customary post-closing adjustments. See the discussion of related pro forma adjustments within the paragraph below. The information and exhibits contained in the Company's Current Report on Form 8-K filed with the Securities and Exchange Commission on June 20, 2025 are incorporated by reference into this Exhibit 99.2.

The Unaudited Pro Forma Condensed Consolidated Financial Statements presented below have been derived from the Company's historical consolidated financial statements and give pro forma effect to Transactions A, B and C (the "Transactions"). The Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2025 reflects the Company's financial position as if the Transactions had occurred on September 30, 2025. The adjustments in the "Transaction Accounting Adjustments" columns in the Unaudited Pro Forma Condensed Consolidated Balance Sheet give effect to Transaction A as if it had occurred as of September 30, 2025. The Unaudited Pro Forma Condensed Consolidated Statements of Operations for the nine months ended September 30, 2025 and each of the years ended December 31, 2024 and 2023 reflect the results of operations as if the Transactions had occurred on January 1, 2023 in that they reflect the reclassification of the Company's engineering, hydraulic fracturing and T&D businesses as discontinued operations for all periods presented.

The Unaudited Pro Forma Condensed Consolidated Financial Statements presented below have been derived from, and should be read in conjunction with, the Company's unaudited condensed consolidated financial statements and the notes thereto as of September 30, 2025, and for the nine months ended September 30, 2025, the Company's audited consolidated financial statements and the notes thereto as of December 31, 2024, and for the two years ended December 31, 2024, Management's Discussion and Analysis included in the Company's Quarterly Report on Form 10-Q for the nine months ended September 30, 2025 and Management's Discussion and Analysis included in the Company's Annual Report on Form 10-K for the year ended December 31, 2024. The historical financial results of the engineering, hydraulic fracturing and T&D businesses will be reflected in the Company's consolidated financial statements as discontinued operations under U.S. generally accepted accounting principles ("GAAP") for all periods presented upon the effective date of each transaction.

The Unaudited Pro Forma Condensed Consolidated Financial Statements are presented based on information currently available, subject to the assumptions and adjustments described in the accompanying notes and is not intended to represent what the Company's condensed consolidated balance sheet and statements of operations actually would have been had the Transactions occurred on the dates indicated above. Further, the Unaudited Pro Forma Condensed Consolidated Financial Statements are provided for illustrative and informational purposes only and are not necessarily indicative of the Company's financial position and results of operations for any future period and does not reflect all actions that may be undertaken by the Company following the closing of the Transactions. In addition, the Unaudited Pro Forma Condensed Consolidated Financial Statements do not reflect the realization of any expected cost savings, synergies or dis-synergies as a result of the Transactions. The actual financial position and results of operations may differ significantly from the pro forma amounts reflected herein due to a variety of factors. Management believes these assumptions and adjustments are reasonable, given the information available

at the time of filing. The Unaudited Pro Forma Condensed Consolidated Financial Statements should be read in conjunction with the Company's historical consolidated financial statements and accompanying notes. The Unaudited Pro Forma Condensed Consolidated Financial Statements presented below have been prepared in accordance with Article 11 of Regulation S-X, Pro Forma Financial Information.

The pro forma adjustments are based on currently available information and assumptions management believes are, under the circumstances and given the information available at this time, reasonable, and best reflect the Transactions on the Company's financial condition and results of operations. The adjustments included within the "Discontinued Operations" columns of the Unaudited Pro Forma Condensed Consolidated Financial Statements are the Company's current preliminary estimates on a discontinued operations basis and could change as the Company finalizes discontinued operations accounting to be reported in the Annual Report on Form 10-K for the year ending December 31, 2025.

Unaudited Pro Forma Condensed Consolidated Balance Sheet As of September 30, 2025

ASSETS		listorical s reported)	O	continued perations saction A) (a)	Ac Ad	ansaction counting justments nsaction A)]	Pro Forma
CURRENT ASSETS				(in thousands, e	xcept share	data)		
Cash and cash equivalents	\$	98,167	\$	_	\$	23,502 (b)	\$	121,669
Marketable securities		12,733		_		_		12,733
Restricted cash		29,461		_		2,450 (c)		31,911
Accounts receivable, net		41,893		(2,248)		_		39,645
Inventories		4,269		_		_		4,269
Current assets held for sale		5,618		_		_		5,618
Other current assets		4,310		(402)		_		3,908
Current assets of discontinued operations		2,558		_		_		2,558
Total current assets		199,009		(2,650)		25,952		222,311
Property, plant and equipment, net		83,828		(51)		_		83,777
Sand reserves, net		39,713		`		_		39,713
Operating lease right-of-use assets		3,592		(559)		_		3,033
Other non-current assets		6,219		`		_		6,219
Noncurrent assets of discontinued operations		4,392		_		_		4,392
Total assets	\$	336,753	\$	(3,260)	\$	25,952	\$	359,445
LIABILITIES AND EQUITY								
CURRENT LIABILITIES								
Accounts payable	\$	12,219	\$	(1,107)	\$	_	S	11.112
Accrued expenses and other current liabilities	<u> </u>	17,519		(1,309)		367 (d)	-	16,577
Current liabilities held for sale		42						42
Current operating lease liabilities		2,693		(188)		_		2,505
Income taxes payable		46,634)		574 (e)		47,208
Current liabilities of discontinued operations		1,052		_		_		1,052
Total current liabilities		80,159		(2,604)	_	941		78,496
Deferred income tax liabilities		2,541		_		_		2.541
Long-term operating lease liabilities		2,028		(395)		_		1,633
Asset retirement obligations		2,722		_		_		2,722
Other long-term liabilities		60		(21)		_		39
Noncurrent liabilities of discontinued operations		_				_		_
Total liabilities		87,510		(3,020)		941		85,431
COMMITMENTS AND CONTINGENCIES								
DALIGNA								
EQUITY								
Equity:		482		_				482
Common stock, \$0.01 par value Additional paid-in capital		540,842		_		_		540,842
Accumulated deficit		(287,947)		(240)		25,011 (f)		(263,176)
				(240)		23,011 (f)		
Accumulated other comprehensive loss		(4,134)		(240)		25.011		(4,134)
Total equity	Φ.	249,243	6	(240)	•	25,011	\$	274,014
Total liabilities and equity	\$	336,753	\$	(3,260)	\$	25,952	2	359,445

Unaudited Pro Forma Condensed Consolidated Statement of Operations For the Nine Months Ended September 30, 2025

	Historical (as reported	. (Discontinued Operations (Transaction A) (a)		Pro Forma			
	(in thousands, except per share amounts)							
REVENUE								
Services revenue	\$ 30	791 \$	(11,967)	\$	18,824			
Services revenue - related parties	1	168	_		1,168			
Product revenue		843			14,843			
Total revenue	46	802	(11,967)		34,835			
COST, EXPENSES AND GAINS								
Services cost of revenue	25	889	(9,010)		16,879			
Services cost of revenue - related parties		288			288			
Product cost of revenue	14	886	_		14,886			
Selling, general and administrative	14	992	(1,170)		13,822			
Depreciation, depletion, amortization and accretion	7	679	(18)		7,661			
Gains on disposal of assets, net	(2	675)	_		(2,675)			
Impairment of long-lived assets	31	669	_		31,669			
Total cost, expenses and gains, net	92	728	(10,198)		82,530			
Operating income (loss)	(45)	926)	(1,769)		(47,695)			
OTHER INCOME (EXPENSE)								
Interest income (expense and financing charges), net	1	434	(322)		1,112			
Interest expense and financing charges, net - related parties		_	_		_			
Other expense, net	(2	791)	(1)		(2,792)			
Total other (expense) income, net	(1	357)	(323)		(1,680)			
Loss before income taxes	(47)	283)	(2,092)		(49,375)			
Provision for income taxes	2	044	(1)		2,043			
Net loss from continuing operations	(49	327)	(2,091)		(51,418)			
Net income from discontinued operations, net of income taxes	45	023	25,011		70,034			
Net (loss) income	\$ (4	304) \$	22,920	\$	18,616			
Net loss per share from continuing operations (basic and diluted)	\$ (1.02)		\$	(1.07)			
Net income per share from discontinued operations (basic and diluted)		0.93			1.45			
Net (loss) income per share (basic and diluted)		0.09)		\$	0.39			
Weighted average number of shares outstanding (basic and diluted)		245		_	48,245			

Unaudited Pro Forma Condensed Consolidated Statement of Operations For the Year Ended December 31, 2024

	Pro Forma (previously adjusted for Transactions B & C)		stments B & C	Discontinued Operations (Transaction A)	Pro Forma (combined)			
	 (in thousands, except per share amounts)							
REVENUE								
Services revenue	\$ 41,486	\$	771	\$ (17,263)	\$	24,994		
Services revenue - related parties	1,548		_	_		1,548		
Product revenue	19,026		31	_		19,057		
Total revenue	62,060		802	(17,263)		45,599		
COST, EXPENSES AND GAINS								
Services cost of revenue	44,789		(7,394)	(13,017)		24,378		
Services cost of revenue - related parties	366		_			366		
Product cost of revenue	18,911		(1,120)	_		17,791		
Selling, general and administrative	116,872		(729)	(1,769)		114,374		
Depreciation, depletion, amortization and accretion	11,985		(241)	(29)		11,715		
(Gains) losses on disposal of assets, net	(3,483)		721	_		(2,762)		
Total cost, expenses and gains	 189,440		(8,763)	(14,815)		165,862		
Operating loss	 (127,380)		9,565	(2,448)		(120,263)		
OTHER INCOME (EXPENSE)								
Interest expense and financing charges, net	(5,254)		470	(6)		(4,790)		
Interest expense and financing charges, net - related parties	(4,707)		_	_		(4,707)		
Other expense, net	(64,565)		1	_		(64,564)		
Total other expense, net	 (74,526)		471	(6)		(74,061)		
Loss before income taxes	(201,906)		10,036	(2,454)	-	(194,324)		
Provision (benefit) for income taxes	(11,294)		2	(13)		(11,305)		
Net loss	\$ (190,612)	\$	10,034	\$ (2,441)	\$	(183,019)		
Net loss per share (basic and diluted)	\$ (3.97)				\$	(3.81)		
Weighted average number of shares outstanding (basic and diluted)	48,065					48,065		

Unaudited Pro Forma Condensed Consolidated Statement of Operations For the Year Ended December 31, 2023

	adjusted fo	Pro Forma (previously adjusted for Transactions B & C)		Adjustments tions B & C	Discontinued Operations (Transaction A)	Pro Forma (combined)	
	(in thousands, except per share amounts)						
REVENUE							
Services revenue	\$	46,996	\$	906	\$ (14,656)	\$	33,246
Services revenue - related parties		980		_	_		980
Product revenue		39,285		(137)			39,148
Total revenue		87,261		769	(14,656)		73,374
COST, EXPENSES AND GAINS							
Services cost of revenue		57,023		(17,669)	(10,977)		28,377
Services cost of revenue - related parties		475		_	_		475
Product cost of revenue		27,489		(1,823)	_		25,666
Selling, general and administrative		29,197		(7,025)	(1,722)		20,450
Depreciation, depletion, amortization and accretion		22,357		(474)	(25)		21,858
(Gains) losses on disposal of assets, net		(5,622)		2,114	_		(3,508)
Impairment of goodwill		1,810		_	_		1,810
Total cost, expenses and gains	·	132,729		(24,877)	(12,724)		95,128
Operating loss		(45,468)		25,646	(1,932)		(21,754)
OTHER INCOME (EXPENSE)							
Interest expense and financing charges, net		(9,997)		2,683	40		(7,274)
Interest expense and financing charges, net - related parties		(1,241)		_	_		(1,241)
Other expense, net		42,845		2	18		42,865
Total other expense, net		31,607		2,685	58		34,350
(Loss) income before income taxes		(13,861)		28,331	(1,874)		12,596
Provision for income taxes		12,199		54			12,253
Net (loss) income	\$	(26,060)	\$	28,277	\$ (1,874)	\$	343
Net (loss) income per share (basic and diluted)	\$	(0.55)				\$	0.01
Weighted average number of shares outstanding (basic and diluted)		47,777					47,777

Notes to Unaudited Pro Forma Condensed Consolidated Financial Statements

The Unaudited Pro Forma Condensed Consolidated Balance Sheet and Unaudited Pro Forma Condensed Consolidated Statements of Operations include the following adjustments:

Engineering Discontinued Operations:

(a) Reflects the discontinued operations of the Engineering Business, including associated assets, liabilities, equity and results of operations. In accordance with ASC 205-20, *Presentation of Financial Statements - Discontinued Operations*, the amounts exclude general corporate overhead costs which were historically allocated, but did not specifically relate to the Engineering Business, as they did not meet the discontinued operations criteria. Such allocations included labor and non-labor expenses related to the Company's corporate support functions (e.g., executive, information technology, human resources, legal, accounting, among others) that historically provided support to the Engineering Business.

Engineering Transaction Accounting Adjustments:

- (b) Reflects the cash proceeds of approximately \$23.5 million received from the buyer from the disposal of the Engineering Business.
- (c) Reflects cash of \$2.5 million deposited into an escrow account by the buyer for the purposes of funding post-closing adjustments and indemnified liabilities.
- (d) Reflects approximately \$0.4 million of costs associated with Transaction A to be incurred subsequent to September 30, 2025.
- (e) Reflects approximately \$0.6 million of estimated current income tax payable associated with the estimated taxable gain from Transaction A. The tax effect of the proforma adjustments was calculated using the historical statutory rates in effect for the period presented.
- (f) Reflects an estimated gain of \$24.8 million related to Transaction A based on the estimate of \$26.0 million of consideration less transaction costs of \$0.4 million, net income tax liabilities of \$0.6 million and the engineering net assets as of September 30, 2025 of \$0.2 million. The actual gain recorded upon close may be subject to change and will be based on amounts as of the close date. Since the Unaudited Pro Forma Condensed Consolidated Statements of Operations only include continuing operations, the estimated gain on sale is not included for the years ended December 31, 2024 and 2023. For the nine months ended September 30, 2025, because discontinued operations are presented in the Company's Quarterly Report on Form 10-Q, the estimated gain on sale is reflected in the pro forma adjustments.

In connection with Transaction A, the Company entered into a transition services agreement with Qualus whereby the MEP will provide certain post-closing services to Qualus on a transitional basis. This agreement is not expected to have a material impact on the periods presented in these Unaudited Pro Forma Condensed Consolidated Financial Statements.

Subsequent Adjustments to Transaction B & C:

Reflects the discontinued operations of Silverback Energy LLC ("Silverback") and Anaconda Manufacturing LLC ("Anaconda"). Silverback and Anaconda provided trucking and manufacturing operations, respectively, primarily in support of the Company's hydraulic fracturing operations. As a result of the Pressure Pumping Transaction, Silverback and Anaconda ceased operations. In accordance with ASC 205-20, *Presentation of Financial Statements – Discontinued Operations*, these entities meet the criteria to be included in discontinued operations. Additionally, post-closing adjustments related to Transaction C are reflected in the pro forma financial information as discussed above.